







**FINAL REPORT** March 2017

# **Recruiting and supporting** care leavers in Australian higher education

**Dr Andrew Harvey Dr Perri Campbell** Lisa Andrewartha Associate Professor Jacqueline Wilson **Pearl Goodwin-Burns** 

#### **ENQUIRIES**

Centre for Higher Education Equity and Diversity Research **E** cheedr@latrobe.edu.au La Trobe University Victoria 3086

**T** +613 9479 5656

latrobe.edu.au/cheedr

Recruiting and supporting care leavers in Australian higher education is published by the Centre for Higher Education Equity and Diversity Research, La Trobe University La Trobe University Melbourne Victoria 3086 Australia Tel: +61 3 9479 5656 Email: <u>cheedr@latrobe.edu.au</u> Web: latrobe.edu.au/cheedr

March 2017

ISBN: 978-0-9946100-4-1

© Centre for Higher Education Equity and Diversity Research, La Trobe University 2017

Written by Dr Andrew Harvey, Dr Perri Campbell, Lisa Andrewartha, Associate Professor Jacqueline Wilson, Pearl Goodwin-Burns.

**To cite this article:** Harvey, A., Campbell, P., Andrewartha, L., Wilson, J., Goodwin-Burns, P. (2017). Recruiting and supporting care leavers in Australian higher education. Report for the Australian Government Department of Education and Training.

© Centre for Higher Education Equity and Diversity Research, La Trobe University 2017

Except as permitted under the Copyright Act 1968 (Australia), this publication may not be reproduced, stored in a retrieval system, communicated or transmitted, in any form or by any means, without the prior written permission of La Trobe University. All enquiries should be made to the address set out on page one of this publication.

The information contained in this publication is of a general nature only. While the University attempts to ensure information is current at the time of publication, it gives no warranties including as to the accuracy or completeness of any such information or of any link or reference contained in this document. The University reserves the right to make any changes to this publication at any time without notice.

# Acknowledgements

The authors acknowledge the funding of the Australian Government Department of Education and Training through the Higher Education Participation and Partnerships Program (HEPPP) - National Priorities Pool 2015.

The original Chief Investigators were:

- Dr Andrew Harvey, Director, Centre for Higher Education Equity and Diversity Research, La Trobe University
- Lisa Andrewartha, Senior Research Officer and Senior Project Coordinator, Centre for Higher Education Equity and Diversity Research, La Trobe University
- Associate Professor Jacqueline Wilson, Associate Professor, Faculty of Education and Arts, Federation University Australia
- Alan Beckley, Adjunct Fellow, Dean's Unit School of Social Sciences & Psychology, Western Sydney University
- Gabrielle O'Brien, (formerly) Senior Equity Officer, Queensland University of Technology and President, Equity Practitioners in Higher Education Australasia (EPHEA).

Other members of the project team included:

La Trobe University

- Dr Perri Campbell, Senior Research Officer, Centre for Higher Education Equity and Diversity Research
- Michael Luckman, Senior Research Officer, Centre for Higher Education Equity and Diversity Research
- Rachael Spicer, (formerly) Manager, Access and Achievement Research Unit.

Federation University of Australia

Pearl Goodwin-Burns, Care Leaver Coordinator

Queensland University of Technology

• Kate Flynn, Communications and Projects Officer, Equity Services

Western Sydney University

- Jim Micsko, Manager, Widening Participation
- Emma Dart (formerly), Care Co-ordinator, Office of Widening Participation
- Genevieve Mosely, (formerly) Office of Widening Participation
- Associate Professor Susanne Gannon, Equity strand leader in the Centre for Educational Research.

We are grateful for the additional advice and assistance provided by:

- Frank Golding, Vice President, Care Leavers Australasia Network (CLAN)
- Richard Pourau, Project Assistant, Western Sydney University
- Shioban Downing, Community Engagement Coordinator, La Trobe University.

To cite this article: **Harvey, A.,** Campbell, P., Andrewartha, L., Wilson, J., Goodwin-Burns, P. (2017). Recruiting and supporting care leavers in Australian higher education. Report for the Australian Government Department of Education and Training.

# Contents

Acknowledgements
Contents
Executive Summary
Project background and report structure
Section 1. Care leavers in higher education: lessons from international and national research
United Kingdom9
Continental Europe
United States
Australia11
Section summary14
Section 2. Methods for collecting data on care leavers in higher education
Educational Access Schemes
Enrolment forms
Bursary applications17
Sample data17
Section summary19
Section 3. The voice of care leavers: findings from in-depth interviews
Participant care experiences
Thematic analysis of interview data24
Section summary
Section 4. Higher education guidelines for professional and academic staff working with care leavers
Section 5. Higher education guidelines for students who spent time in out-of-home care
References
Appendix: Interview guide

## **Executive Summary**

This report follows a multi-state, cross-institutional analysis of care leavers in Australian higher education. Those who have left out-of-home care – including foster, residential and kinship care – typically face extreme challenges to participate and succeed in higher education, highlighted both by our multi-state interviews and our national data analysis. Care leavers, for example, are relatively likely to suffer from anxiety and/or mental health disorders, to be from low socio-economic status backgrounds, and to have suffered disrupted schooling and family lives. While some of these issues are beyond the scope of universities to address, there are numerous policy changes that would improve access and success. Moreover, our evidence also highlights that care leavers can bring particular strengths and forms of capital to higher education which not only help them to succeed, but which could be harnessed more broadly by institutions. Transitioning from a deficit model to a richer understanding of both the challenges and strengths of care leavers has the potential to benefit all students on campus.

International evidence is clear on the extent of under-representation of care leavers in higher education, but also on the potential for growth. Our review of recent research confirms that policy changes at both government and institutional level can dramatically improve access and success rates. Moreover, since our initial report (Harvey, McNamara, Andrewartha, & Luckman, 2015), several more dedicated educational resources have been developed internationally to support schools and universities, such as a practical guide to educating children and young people in care in the United Kingdom (Cameron, Connelly, & Jackson, 2015). Research continues to confirm the importance of raising expectations, normalising and demystifying higher education as a destination, listening to the voices of care leavers, providing targeted and accessible support, and measuring university participation, success and outcomes.

In this project, our collection of data and institutional advocacy enabled development of important new methods of measuring access and success. Through advocacy of the research team, changes were made to university enrolment forms, and the educational access schemes operated by state tertiary admissions centres. Introduction of these two methods, as previously advocated in our initial report, now provides a template for collection of data at or before enrolment, by which institutions can quantify and monitor care leaver data. In addition, one partner institution was able to identify care leaver students through a new bursary scheme, providing a further point of data access. Similar to disability, care leaver status may be self-identified subsequent to enrolment, enabling targeted resources to be formed and outcomes to be tracked. Through the various modes of data collection, we have the first understanding of likely care leaver enrolments in Australian higher education, and templates that can be expanded across the sector to build an evidence base. Further, from the sample of students we were able to monitor, we received some important initial indications of course profile and achievement. Perhaps most notable here was the relatively high achievement of enrolled care leavers, the reasons for which became increasingly clear from our qualitative research.

Our qualitative research involved detailed interviews with university care leavers across three states and four institutions. Several themes were clear, including the desire of care leavers to be treated as normal students, an over-representation of anxiety and mental health issues, a relative unlikeliness to report such issues within the institution or to seek help through counselling and other services, a powerful resilience and determination to succeed, and potential issues of campus climate where experiences clearly differed from other students involved in coursework and informal conversations. For Indigenous care leavers, specific cultural connections and issues were strong themes, and access to a dedicated Indigenous unit on campus was seen as beneficial not only for direction to other university services, but as a space of cultural understanding and community. To raise university access levels, participants noted the need for further education of carers, case workers, and related staff, and for higher expectations among school teachers and others.

One of the most notable lessons from our research relates to a discrepancy between the genuine disadvantage of many care leavers and their relatively low take-up of financial and support services. Interviews revealed the reality of care leaver experiences, in which financial hardship was common, mental health was frequently cited, and disadvantage was often extreme. Our quantitative analysis, however, suggests that eligible care leavers rarely apply for compensation for educational disadvantage through the tertiary admissions centre schemes, rarely apply for scholarships and financial bursaries, and often do not access disability, counselling and other support services once enrolled. The gap between lived experience and request for support highlights the need to normalise student support and service use, and to simplify and demystify bursary and scholarships applications. Research suggests that care leavers are particularly likely to be confused by tertiary application processes, including financial aid, and reluctant to seek services and deal with bureaucracies, particularly where privacy fears are involved. Mitigating these concerns is important not only to ensure that care leavers are receiving compensation that is deserved, but to ensure that other disadvantaged students are not similarly being marginalised by systems that reward cultural capital and inside knowledge rather than genuine disadvantage.

Equally, the qualitative research and quantitative data provided corroborated evidence of resilience and determination among care leavers. This resilience has arguably been understated by previous research focussed on barriers and disadvantage. However, the rapid rise in British university enrolments of care leavers since the adoption of targeted policies and strategies in 2004 – from 1% to 7% - provided early evidence of the potential for improvement. One of the reasons such enrolment growth was possible was that care leavers often have reserves of determination and persistence that enable them to overcome severe educational barriers and challenges. Our quantitative data revealed relatively strong academic performance, and our qualitative research also highlighted uncommon levels of persistence and independence among care leavers. These strengths reveal the capacity for similar growth in access and success if Australian policies were adapted to reflect international best practice. Interviews also revealed the need for further research into the qualities and assets that care leavers, and other under-represented students, can bring to higher education. Relatedly, some interview participants observed issues of campus climate, for example perceiving that other students had limited life experience and focussed on issues that appeared trivial or inappropriate to the care leaver cohort. Wider challenges exist for universities to promote inclusive excellence and to understand the potential benefits of diversity for student learning and growth. There also remains a challenge to translate academic success to retention and completion, as financial and other barriers often remain prevalent throughout the candidature of care leavers.

Our report concludes with guidelines for prospective care leaver students, and a set of guidelines for university staff. The student guidelines address many of the issues raised above, including demystifying higher education and providing practical guidance around course and university selection, enrolment, and access to bursaries, scholarships, and services. The guidelines for staff focus on the need to understand specific concerns of care leavers, but also to normalise higher education and to keep expectations of achievement high, particularly in outreach activities. Students interviewed highlighted university outreach as central to raising participation, and it is important that universities conduct their outreach activities not only with proximate secondary schools, but with flexible learning schools and other educational sites, working in conjunction with welfare agencies and other service providers where relevant.

# Project background and report structure

#### Background

Out-of-home care refers to the placement of children who are unable to live with their families with alternate carers, often due to child abuse and neglect (Victorian Department of Human Services, 2007). Types of out-of-home care include foster care, relative/kinship care, residential care, family group homes, and independent living arrangements (Australian Institute of Health & Welfare [AIHW], 2016). Approximately 43,400 children and young people live in out-of-home care in Australia, and a substantial proportion of these young people come from low socio-economic status, regional, and/or Indigenous backgrounds (AIHW, 2016; Harvey, McNamara, Andrewartha & Luckman, 2015; State Government of Victoria, 2012). Families faced with financial hardship are more frequently challenged by mental health issues, misuse of alcohol and other drugs, violence, unemployment, and housing insecurity (Australian Council of Social Services, 2014; McLachlan, Gilfillan, & Gordon, 2013). These pressures can increase the likelihood of neglect and abuse leading to children being placed in out-of-home care (Australian Institute of Family Studies, 2015; Garbarino, 1977). Young people in out-of-home care confront specific educational challenges from an early age, including placement instability and disrupted schooling (Bromfield, Higgins, Osborn, Panozzo, & Richardson, 2005; CREATE Foundation, 2006; Fernandez, 2008; Townsend, 2012). The relatively low school achievement and completion rates of young people in out-of-home care are well documented (AIHW, 2007, 2011).

The term 'care leavers' can be used to refer to people who spent time in out-of-home care before 18 years of age and subsequently transitioned out of the system (Harvey, McNamara, Andrewartha & Luckman, 2015). Legal protection and formal assistance from state governments typically decrease or discontinue entirely at 18 years of age (Creed et al., 2011; Victorian Department of Human Services, 2007; Mendes, 2009: 34). Care leavers are at increased risk of homelessness, unemployment, and dependence on Centrelink (Johnson et al., 2010; McDowall, 2013; Thoresen & Liddiard, 2011). No robust data are available on the number of care leavers in higher education in Australia. Available evidence, however, suggests that Australian care leavers rarely transition to higher education (Cashmore, Paxman, and Townsend, 2007; Harvey et al., 2015; McDowall, 2009; Mendes, Michell, & Wilson, 2014).

Despite the severe educational and broader challenges faced by many care leavers, evidence also suggests that their participation in higher education can be increased rapidly through targeted policies and commitments. Section 1 highlights how governments and universities within the United Kingdom, in particular, have introduced policies which have led to rising access and participation. Moreover, many care leavers in higher education record strong achievement levels and successful graduate outcomes. While still inchoate, international evidence suggests that care leavers may hold relatively high levels of resilience, determination, adaptability, and other qualities that are undervalued and under-researched in higher education. Understanding the different forms of 'capital' that students possess is crucial to increasing campus diversity (Yosso, 2005), and ensuring that all students are respected and understood for their strengths. In exploring the specific strengths of care leavers in higher education, we can therefore understand not only how policies need to change, but why that change is likely to be effective both in raising participation and in increasing campus diversity and quality.

The current project, *Recruiting and supporting care leavers in Australian higher education*, builds on an initial national project conducted by La Trobe University in 2015, and funded through National Centre Student Equity in Higher Education at Curtin University (Harvey et al., 2015). That project,

*Out of care, into university,* was the first national project to map the higher education sector in relation to care leavers. The project included an examination of relevant education and child protection data sources; a survey of public universities; and interviews with community service organisations. Findings revealed that existing data on the education of Australians in out-of-home care was limited, and there were few policies, procedures, or programs catering to care leavers in higher education. Included in the project report were 26 recommendations targeted to the Australian Government, state and territory governments, higher education institutions, and community service organisations. Our research from *Out of care, into university* also provided an evidential foundation for the current project.

Research for the current project was led by La Trobe University in partnership with Federation University of Australia, Queensland University of Technology, and Western Sydney University. The project was funded through an external research grant provided by the Australian Government Department of Education and Training through the Higher Education Participation and Partnerships Programme - National Priorities Pool 2015. Two primary research questions were addressed: (1) How can universities increase access to higher education for care leavers? (2) How can universities support care leavers who are studying at university?

The five main components of the project were: (1) a review of international research, policy, and legislation relating to care leavers in higher education; (2) development of methods for higher education institutions to collect data on care leaver students; (3) in-depth interviews with care leavers in higher education about their views and experiences; (4) development of guidelines for people from out-of-home care backgrounds considering higher education; (5) development of guidelines for guidelines for university staff delivering outreach to young people in out-of-home care and providing support for care leavers in higher education.

#### Report structure

Section one of this report provides an overview of recent international and national research, policy, and legislation on supporting care leavers in higher education.

Section two outlines methods for collecting data on care leavers across higher education institutions, and provides a summary of data collected within the project.

Section three provides an analysis of interviews with care leavers who are enrolled at university, and captures their experiences accessing and participating in higher education.

Section four provides guidelines for professional and academic staff working with care leavers in higher education.

Section five provides guidelines about accessing higher education for students who spent time in out-of-home care.

# Section 1. Care leavers in higher education: lessons from international and national research

This section of the report provides a summary of recent international and national research, policy, and legislation on supporting care leavers in higher education.

## United Kingdom

The first major research project to examine the experiences of care leavers in higher education in the United Kingdom (UK) was the *By Degrees: Going to University from Care* project (Jackson, Ajayi, & Quigley, 2005). The researchers followed 50 care leaver university students per year over three years. Care leavers reported a lack of information and advice about universities and courses and uncertainty about the financial and accommodation support available. Students staying in distant accommodation had limited social relationships and participation in university life, and students without adequate financial support often spent considerable time in paid work. At the time of the report, only one university in the UK had a comprehensive care leaver policy. The researchers recommended that 'All higher education institutions should have a comprehensive policy for recruitment, retention and support of students from a care background' (p. xiv).

Since the *By Degrees* report, considerable progress has been made within British higher education sector. Care leavers are formally recognized as an under-represented group in higher education and their participation is closely monitored in England, Northern Ireland, Scotland, and Wales. Buttle UK, a trust that supports children and young people in need, has also played an important role in supporting care leavers in higher education. From 2006, Buttle UK began awarding a 'Quality Mark' to higher education providers that demonstrated commitment to care leavers. This accreditation system helped to drive sectoral change, and successful universities have introduced a range of initiatives including: prioritising care leavers in outreach programmes and other university activities; recruiting designated staff members to support care leaver students at pre-entry and post-enrolment; offering bursaries and hardship funds for care leavers; and giving preferential access for care leavers to schemes to support disadvantaged students (Rawson, 2016, p. 5). The Quality Mark, which operated between 2006 and 2015, has now been discontinued, with a shift towards embedding support for care leavers into mainstream provision (Rawson, 2016; p. 2).

England has maintained a strong focus on the higher education participation of care leavers for over a decade. In 2006, England's Office for Fair Access (OFFA) wrote to all institutions with access agreements to encourage them to address the needs of care leavers in their access agreements. There have also been several additions to the legislation on the education of care leavers in England and Wales. The *Children (Leaving Care) Act 2000* introduced the first statutory requirement for local authorities to support young people aged 16 to 24 in education. The *Children and Young Persons Act 2008* established a statutory £2,000 local authority bursary for young care leavers at university. From April 2011, *The Children Act 1989 Guidance and Regulations Volume 3: Planning Transition to Adulthood for Care Leavers* implemented a suite of regulations and guidance around educational transitions (All-Party Parliamentary Group for Looked After Children and Care Leavers, 2012). Data on the care leaver status of students is now officially collected by the Higher Education Statistics Authority (HESA), which means that all higher education institutions can identify care leavers within their institutions. OFFA is also able to monitor improvements in the proportion of care leavers reaching university (OFFA, 2015). Along with the above initiatives, the National Network for the Education of Care Leavers (NNECL) was formed in 2013 to support care leavers into higher education in England (NNECL, 2013). In partnership with higher education providers, the NNECL developed a guide for supporting care leavers in further education and higher education. The main components related to: accommodation support; financial support; designated contact people; and identification of care leavers via a tick box at enrolment.

Evidence from the UK shows that some care leavers are reluctant to disclose their care leaver status to fellow students, support staff, or academics for fear of prejudice (Jackson & Cameron, 2014). Some types of support can be offered while maintaining confidentiality, such as tuition fee-waivers, residential scholarships, and social and academic support provided online. Group identity can also be framed as a positive force. A UK website called Propel was created by the Who Cares? Trust which encourages care leavers to inform universities of their status in order to access extra information, guidance, and support (Who Cares? Trust, 2016).

The combined policy and legislative focus on care leavers has increased the proportion of care leavers continuing to higher education. In England, this proportion increased from approximately 1 per cent of 19 year old care leavers in 2003 (Department for Business Innovation & Skills, 2014) to 6-7 per cent of 19-21 year old care leavers in 2014 (Department of Education, 2014). While this figure remains well below the 40 per cent higher education participation rate of young people in the general population, it demonstrates significant improvement over time (The Centre for Social Justice, 2015). As our subsequent interviews and quantitative data reveal, rapid enrolment growth is likely to be partially linked to the underestimated capacity of care leavers to overcome educational disadvantage through determination and resilience.

## Continental Europe

Following the *By Degrees* research project, the *Young people from a public care background pathways to education in Europe* research project (YiPPEE) examined the education pathways of care leavers across England, Denmark, Sweden, Hungary, and Spain. The researchers found that people from public care experienced severe educational disadvantage across all five countries. For example, young people from care were: 'under pressure to opt for short-cycle occupational training in order to become economically independent as soon as possible rather than higher level academic or vocational options with the potential to lead to more satisfying careers in the longer run' (Jackson & Cameron, 2012, p. 8). Factors that facilitated success in education included: stability of placement and schooling; placement with carers who gave priority to education; and sufficient financial support and suitable accommodation. The research also highlighted that 'reliable statistical information is an essential basis for improving the educational opportunities of young people who have been in care' (Jackson & Cameron, 2012, p. 10). Jackson and Cameron (2014) published a comprehensive account of their cross-nation findings in their subsequent book, Improving Access to Further and Higher Education for Young People in Public Care.

## United States

There has been a considerable amount of research focussing on the college experiences of people from foster care in the United States. Wolanin (2005) reported that young people in foster care were less likely to complete high school than their peers (50 per cent and 70 per cent, respectively). Those who completed high school were also less likely to enrol in higher education than their peers (20 per cent and 60 per cent, respectively). Barriers to college access included: inability to meet admission standards; a belief that college is 'not for people like me'; and financial constraints.

Three significant federal laws passed that were designed to increase college access for people from foster care in 2008: the *Fostering Connections Act; College Cost Reduction Act;* and the *Higher Education Opportunity Act* (Legal Center for Foster Care and Education, 2008). Some US states have

extended foster care beyond 18 years of age and this change has been associated with increased participation in higher education (Courtney & Dworsky, 2005). In the US, the ability to remain in foster care beyond 18 years of age differs between states. Courtney and Dworsky (2005) compared the educational outcomes of young people from Illinois, who are allowed to remain in care until they are 21, to the outcomes of young people from Iowa and Wisconsin, who typically age out of foster care at 18 years of age. Young people who were still in care at age 19 were more than twice as likely to be enrolled in a school or training program as those who had been discharged from care.

Research from the US has found that people from foster care who do access college also have lower college completion rates than their peers (Courtney, Dworsky, Lee & Rap, 2010; Courtney et al., 201; Pecora et al., 2003; 2005). Courtney et al. (2011), for example, found that only four per cent of students from foster care had completed a four-year college degree by the age of 26, compared to 36 per cent of the general population. Success at college is impeded by: inadequate information; lack of family support; no stable home base, and inadequate support though financial aid, student services, and counselling.

There are several examples from the US of targeted and multi-pronged strategies that have proven successful for students from foster care. The Blavin Scholars Program at the University of Michigan, for example, offers academic guidance and mentorships, along with scholarships, housing, and other additional resources for youth aging out of foster care who wish to pursue post-secondary education (Blavin Scholars, 2016). The program has a 95 per cent retention and graduation rate. Better Futures model is another example from the United States where coaching and mentoring from older peers with foster care experience has been shown to promote acceptance and academic confidence (Phillips et al., 2015).

## Australia

Approximately 43,400 children live in out-of-home care in Australia and this number has been increasing for over a decade (Australian Institute of Health and Welfare (AIHW), 2016). Approximately 93 per cent of children are in out-of-home care on care and protection orders issued by a statutory authority or court (AIHW, 2016). Once a child has been placed in care, the state government acts as a corporate parent and assumes responsibility for decision-making and care arrangements. Typically, children in care confront educational challenges and disruptions from an early age. A substantial proportion of these children come from low socio-economic status, regional, and/or Indigenous backgrounds (AIHW, 2016; Harvey, McNamara, Andrewartha & Luckman, 2015; State Government of Victoria, 2012). Aboriginal and Torres Strait Islander children are particularly over-represented in the child welfare system. Aboriginal and Torres Strait Islander children are approximately seven times as likely as non-Indigenous children to be receiving child protection services (AIHW, 2016).

Australian research has consistently documented the relatively poor overall school achievement and school completion rates of young people in out-of-home care. Factors that mitigate school success include: placement instability; frequent school changes; absenteeism; bullying; lack of family support; trauma associated with past abuse and neglect; mental health issues; and behavioural issues; lower expectations from carers and teachers (CREATE Foundation, 2006; Creed, Tilbury, Buys, & Crawford, 2011; Fernandez, 2008; Frederick & Goddard, 2010; McFarlane, 2010; Townsend, 2012). Studies by the Australian Institute of Health and Welfare (AIHW) have matched educational achievement data and community service data across multiple jurisdictions (AIHW, 2007, 2011). Results showed that children on guardianship or custody orders, many of whom were in out-of-home care, had poorer reading and numeracy test scores compared with their peers. Furthermore,

Indigenous children on guardianship or custody orders were less likely to achieve the reading and numeracy benchmarks than other children on guardianship or custody orders (AIHW, 2007, 2013).

Australians from out-of-home care face another set of educational barriers when they age out of care. Care leavers can be broadly defined as people who spent time in out-of-home care before they were 18 and have subsequently transitioned out of the system (Harvey, McNamara, Andrewartha & Luckman, 2015). Legal protection and formal assistance from the State government decreases at 18 years of age, with some people leaving care as young as 16 years of age (Creed et al., 2011; Victorian Department of Human Services, 2007; Mendes, 2009: 34). Leaving care often coincides with the final years of secondary school which can have a strong impact educational pathways. After the age of 16, students may elect to discontinue secondary school and enter the workforce or enrol in vocational education to obtain workplace-specific skills and knowledge. In the final years of secondary school, students obtain their Australian Tertiary Admissions Rank (ATAR) or, in Queensland, the 'Overall Position'. The ATAR or Overall Position is used to determine entry into most higher education institutions in Australia (Gale & Parker, 2013).

The abrupt reduction in support that care leavers experience at age 18 contrasts to the experience of most young people in the general population who stay in the parental home receiving continuous support into their early-to-mid 20s (Australian Bureau of Statistics, 2013). Care leavers have very high rates of homelessness, unemployment, dependence on Centrelink, and relatively poor educational outcomes (Johnson et al., 2010; McDowall, 2013; Thoresen & Liddiard, 2011). In recent years, most states and territories have introduced legislation or policy to assist care leavers beyond 18 years of age (Mendes, 2014). However, there are differences across jurisdictions in support types and upper age limits (Department of Social Services, 2014). Under the National Framework for Protecting Australia's Children 2009 – 2020, governments are establishing a nationally consistent approach to supporting care leavers (Department of Social Services, 2014). A landmark study by Deloitte Access Economics (2016), commissioned by Anglicare Victoria, found that extending care to 21 years would halve homelessness, improve mental health outcomes, and double education participation rates. A national campaign led to Anglicare, the Home Stretch campaign, was launched in 2016 to call for governments to extend the leaving care age to 21 years. Extending care would require governments to provide ongoing carer reimbursements to carers, case management to the young person, and resources to access education or employment activity (Home Stretch, 2016).

The limited amount of data available suggest that Australian care leavers rarely transition to higher education (Mendes, Michell, & Wilson, 2014; McDowall, 2009). Cashmore, Paxman, and Townsend (2007) examined the employment and educational outcomes of 47 care leavers in New South Wales. Care leavers were less likely to have completed secondary school compared with same-age counterparts in the general population and less likely to be in full-time work or education four to five years after leaving care. While no national data are collected, a 2009 report by the CREATE Foundation detailed the results of a survey of 471 young people who were in care or had left care. Of those who had left care, only 35 per cent of the sample had completed year 12. At the time of the study, only 11 per cent of the sample were studying at TAFE and 2.8 per cent were studying at university (McDowall, 2009). Mendes et al. (2014) estimated that only one per cent of Australian care leavers transition into higher education, compared to 26 per cent of young people in the general population.

Until recently, few Australian studies had explored the experiences of care leavers in higher education. As a result, 'we know strikingly little about the experiences of care leavers who enter higher education in Australia' (Mendes, Michell, & Wilson, 2014, p. 249). Jurczyszyn and Tilbury (2012) interviewed 13 young people who were in care or leaving care in Queensland. Interest in higher or further education was facilitated by: having a person who helped explore careers and university life; receiving advocacy to overcome practical barriers; and carers and teachers with high

expectations. Mendis, Gardner, and Lehmann (2014) interviewed 18 university-educated women who had spent time in out-of-home care. Experiences differed depending on their personality, resilience, and individual care circumstances. Michell, Jackson, and Tonkin (2015) also published the stories of 14 care leavers who had transitioned to higher education, overcoming obstacles to succeed 'against the odds'. Wilson and Golding (2016) published detailed case studies describing their own 'unexpected, unpredictable and strange' paths from care to university.

The first major national project to map the higher education sector in relation to care leavers was titled *Out of care, into university* (Harvey et al., 2015). The project was conducted by La Trobe University and included an examination of relevant education and child protection data sources; a survey of public universities, and interviews with community service organisations. The project found that existing data on the education of Australians in out-of-home care was limited, and there were few policies, procedures, or programs catering to care leavers in higher education. Only one of the 28 universities surveyed as part of the project collected any data on care leaver students. This university recorded how many students who participated in widening participation chose to disclose their care background.

There is no national method for identifying care leavers within the Australian higher education system. In contrast to England, care leavers do not constitute a distinct equity group in Australian higher education. Six equity groups receive targeted support and funding in Australian higher education. These groups are: people from low socio-economic status (SES) backgrounds; Aboriginal and Torres Strait Islanders; women in non-traditional courses and postgraduate study; people from non-English speaking backgrounds; people with disabilities; and people from rural and isolated areas (Department of Education Employment and Training (DEET), 1990). Higher education data are collected on the equity groups and their access, participation, retention, and completion rates are monitored. A significant number of care leavers fall within the established equity groups, especially students from low socio-economic status, rural, and Indigenous backgrounds (AIHW, 2016; Harvey et al., 2015; State Government of Victoria, 2012). There are currently no methods, however, for separately monitoring small, highly disadvantaged groups such as care leavers on a national level (Harvey, Andrewartha, & McNamara, 2015).

Compared with the UK and the US, Australia has progressed slowly in providing targeted support for care leavers in higher education. Harvey et al. (2015) conducted a national survey of public universities in 2014 and found that few institutional policies and programs existed to improve access and achievement of care leavers. There was often an assumption that care leavers would be included within policies and programs that had been developed for other equity groups, such as students from low socio-economic status backgrounds. The researchers concluded, however, that 'the levels of access for care leavers are so low that direct and explicit support is required to improve their access and recruitment, at both national and institutional levels' (p. 44).

Over the past few years, a small number of Australian universities have begun to introduce targeted initiatives to support care leavers in higher education. These initiatives include: scholarships or bursaries specifically for care leavers (e.g. Federation University Australia, La Trobe University, Swinburne University, Western Sydney University); designated care coordinator positions to support care leavers (e.g. Federation University); and targeted outreach programs for school aged children in out-of-home care (e.g. La Trobe University, Western Sydney University). From 2015, students applying for scholarships through the Victorian Tertiary Admissions Centre have been able to explicitly self-identify as care leavers. Separately, some states have introduced fee waivers for care leavers wishing to pursue vocational education and training at Technical and Further Education Institutions (TAFE) institutes (e.g. South Australia, Victoria, and Western Australia) (Beauchamp, 2014).

### Section summary

Access to higher education for people from out-of-home care backgrounds is a global concern. In the United Kingdom, targeted research led to policy advancement which has significantly increased higher education access and support for care leavers, and which has also influenced broader European policies. Universities have responded to research and policy drivers to include care leavers within their access agreements, and to monitor participation and outcomes for accreditation and evaluation purposes. Handbooks and guidelines for university staff have begun to be introduced.

In the United States, many policy advocates have focussed on extending foster care beyond 18 years of age in an effort to improve transition into post-secondary education. Multi-state research has revealed the influence of extending state care on university participation in particular. Highly focussed support programs at some universities have seen high success rates, but overall retention and completion rates for care leaver students remain relatively low. In Australia, some recent institutional and policy changes have begun to formalise support for care leavers in vocational and higher education, and to enable collection of formal data on access and participation. However, approaches remain confined to institutions or states, and a national approach to attracting, supporting and monitoring care leavers remains elusive.

# Section 2. Methods for collecting data on care leavers in higher education

This section of the report details several different methods of collecting data on care leaver students which were developed and successfully introduced during the course of the project. We have argued previously that the collection and publication of accurate data on care leaver outcomes is central to developing national policy. One of the major findings of the final report from the *Young people from a public care background pathways to education in Europe* (YiPPEE) research project, which examined post care educational outcomes in a number of European countries, was that 'reliable statistical information is an essential basis for improving the educational opportunities of young people who have been in care' (Jackson & Cameron, 2012, p. 10).

Harvey et al. (2015) explored the landscape of data available on care leaver outcomes at national, state, and institutional level and found it to be limited. At the time of the report, there was no systematic collection of national data on care leavers in higher education. State-based child protection departments recorded some details regarding the outcomes of their clients, often through case notes. Some state level data were also collected via surveys. For example, the On Track survey in Victoria (Victoria State Government, 2015) and the Next Step survey in Queensland track the education outcomes of individuals 6 months after leaving school (Queensland Government, 2015). These surveys were administered through the state-based departments of education and included an indicator which denoted whether a survey respondent had been in care. Only aggregate data on the findings of these surveys are publicly available, however, and not the specific outcomes of care leavers. A longer term study is also being managed by the Victorian Department of Human Services, entitled 'Beyond 18: The Longitudinal Study on Leaving Care'. The study includes a survey to understand care leaver experiences with: finding accommodation; finding a job/further education; building a supportive social network; and accessing support services (Australian Institute of Family Studies, 2017). Harvey et al. (2015) conducted a national survey of public universities and found that only one of the 28 universities surveyed as part of the project was formally collecting any data on care leaver students. This university recorded how many of the students who participated in widening participation chose to disclose their care background.

Given the landscape outlined in the 2015 report, the project team expected to confront numerous barriers to the collection and sharing of data at institutional and sectoral level. Many of these barriers were indeed encountered, including difficulty in coordinating multiple university application systems and sub-systems which are controlled by separate stakeholders; the struggle to allocate additional resources and time to modify online systems; and concerns around privacy and confidentiality. Nevertheless, the advocacy of the project team resulted in changes to several institutional and tertiary admission centre processes that will provide baseline care leaver data for the first time, providing momentum for broader sectoral change. In addition, some real-time institutional and state level data was able to be collected, enabling initial reflections on the geodemographics, application rates, academic success, and service take-up, of prospective and currently enrolled care leaver students. Notably, many of the templates and processes developed could also be modified to monitor access of other under-represented groups in higher education, such as carers, single parents, incarcerated people, military veterans, and students from refugee backgrounds.

## Educational Access Schemes

The project team devised a method of identifying care leavers through educational access schemes which are managed by state-based tertiary admissions centres. Educational access schemes, also called special entry access schemes in some states, enable applicants to receive bonus entry points to compensate for educational disadvantage (Queensland Tertiary Admissions Centre, 2017; Universities Admissions Centre, 2017; Victorian Tertiary Admissions Centre, 2017). The Victorian Tertiary Admissions Centre (VTAC) manages the Victorian Special Entry Access Scheme (SEAS). Category 2 of SEAS covers 'difficult circumstances' and allows students to document personal circumstances and experiences that may have negatively affected their educational outcomes. An unpublished examination of SEAS data at La Trobe University from 2011 and 2012 found that only 32 out of 20,000 SEAS applicants had mentioned care experience in their impact statements. While restricted to only one university, this analysis suggested that few care leavers were self-identifying through SEAS in order to receive compensation for educational disadvantage.

Throughout 2016, the project team liaised with VTAC and successfully advocated for sub-questions to be added to Category 2 of SEAS to cater to care leavers. The 2017 VTAC application form was subsequently amended to include the following question: 'Do your difficult circumstances relate to time spent in formal out-of-home care? (e.g. foster care, residential care, kinship care or having been a ward of the state)'. Applicants who answer 'yes' are prompted to specify the type and duration of care as part of their impact statements. A preliminary analysis of La Trobe University SEAS applications in late 2016 found that 163 SEAS applicants had disclosed their care leaver status. This analysis suggests that the additional question has resulted in many more care leavers self-identifying through SEAS, and therefore being potentially eligible to receive compensation. As our interviews of 35 enrolled care leaver students across three states highlights (section 3), care leavers remain relatively unlikely to apply for scholarships, bursaries and compensation through the educational access schemes, despite their eligibility. Changes to the educational access scheme forms could therefore potentially improve financial assistance and participation of care leavers dramatically.

## **Enrolment forms**

The project team also devised a method through which higher education institutions could identify care leaver students at enrolment, similar to the current processes for identifying Indigenous students and students with a disability. After extensive liaison across university departments and campuses, Federation University of Australia was able to introduce a new, mandatory question to identify care leavers. The following question was added to the enrolment checklist for 2017 enrolments:

Have you ever spent time in formal out-of-home care? (Note: out-of-home care could include foster care, residential care, kinship care, or being a ward of the State).

Students can answer this question by selecting Yes/No from a drop down field. As of 3 February 2017, 46 students had ticked 'Yes' to spending time in care through 2017 enrolments. All of these students were new enrolments or students returning from deferral. The project team at La Trobe University has developed a strategy for implementing an equivalent change to enrolment processes in 2017, subject to the approval of senior management. Again, such changes to enrolment forms

enable early identification and the direction of targeted financial support and other resources, while also enabling institutions to monitor progress on access, participation and completion measures. The establishment of a separate enrolment question, including language consistent with the educational access schemes, creates a template for other universities to adopt.

## **Bursary applications**

One of the issues with adding a question on enrolment forms or educational access schemes is the perceived purpose of the question, and there remains a need to link care leaver identification with access to, and provision of, tangible support. To this end, the project team also developed a method of identifying care leavers via targeted bursaries. La Trobe University introduced a new \$500 care leaver bursary system in 2016. The Expression of Interest form for the bursary included the following question:

Please outline the types of formal, statutory out-of-home care you were in, including the length of time in care. Please note: Formal, statutory out-of-home care occurs following a child protection intervention (i.e. through a care and protection court order). It does not cover other out-of-home care situations through alternative arrangements.

Due to the initial low number of care leavers, all domestic, undergraduate care leaver students were eligible to apply for the bursary in 2016. A total of 22 care leaver students were identified through the bursary system.

In 2017, the bursary was restricted to commencing care leaver students. Applicants were also required to provide documentation supporting their time in out-of-home care, such as:

a letter from a responsible person (such as a counsellor, GP, lawyer, teacher, social worker that is aware of your circumstances), documentation from the Department of Human Services, court documents, documentation from your out-of-home care provider or a statutory declaration.

Identification of care leavers not only enables bursaries to be provided, but also enables institutional staff to direct students to other resources, including academic advisers and mentors, residential support, and student services. Our interviews confirmed that for many care leaver students, having access to a central point of contact within a university was highly beneficial. For Indigenous care leavers, the presence of a dedicated Indigenous unit on campus was critical, providing a central liaison point and resource.

## Sample data

In addition to the establishment of data collection frameworks, one of the partner universities was able to obtain aggregate, de-identified data on a cohort of currently enrolled care leaver students. The small sample size (21 students in total) means that any trends must be interpreted cautiously. Nevertheless, the analysis below serves as a model that could be replicated with larger, cross-institutional samples in future, and provides some initial insight into patterns of enrolment. Using the sample data, we compared the geo-demographics, equity group membership, and academic performance of care leaver students with non-care leaver students.

#### Geo-demographics:

Care leaver students were much more likely to be female than the non-care leaver group. In total, 86 per cent of care leaver enrolments were female, compared to 64 per cent of non-care leavers across the university. Care leaver students were also more likely to be older than non-care leavers, with 76 per cent of care leavers over the age of 21, compared to only 45 per cent of non-care leavers. While the overall numbers of care leaver students above the age of 36 were relatively small, they also comprised 19 per cent of the care leaver cohort, compared to only 10 per cent of non-care leavers.

When care leaver enrolments were examined by Broad Field of Education, we found that they were over-represented within the Health Sciences. Forty eight per cent of care leavers were studying a course within the Health Science field compared to only 35 per cent of the non-care leaver group. Care leavers were also over-represented within the Broad Field of Education that comprises Education itself, with 19 per cent of all care leavers studying Education compared to 11 per cent of non-care leavers.

#### Equity:

We also examined the overlap between care leaver status and low socio-economic status, regional status, disability status, Indigenous background, and non-English Speaking Background. Care leavers were much more likely than non-care leavers to come from a low socio-economic status (SES) background. In total, 43 per cent of care leavers were identified as coming from a low SES background, compared to only 20 of the non-care leaver group. Because socio-economic status is defined at a point in time (enrolment) based on geographic location (postcode or census collection district), it may also be that this definition understates the extent of disadvantage among care leavers, as many mature age care leavers are likely to have experienced financial hardship in the near or distant past. It would also be interesting to disaggregate socio-economic status by decile instead of quartile to examine the full extent of socio-economic disadvantage faced by care leaver students.

Care leavers were comparatively over-represented at regional campuses within the university, with 43 per cent of care leavers studying at a regional campus compared to 27 per cent of non-care leavers. Similarly, when examined using the postcode of their permanent residential address, care leavers were more likely to come from regional area, with 48 per cent coming from a regional home postcode compared to 30 per cent of non-care leavers.

Perhaps unsurprisingly, care leavers were much more likely to have formally reported a disability. Over 33 per cent of care leavers reported having a disability at enrolment, compared to only 7 per cent of the non-care leaver cohort. Care leavers are therefore almost five times more likely to have indicated that they have a disability than non-care leavers, and in many cases this is likely to relate to mental health. However, it is also notable that around half of our interviewed care leavers claimed to have suffered from anxiety or mental health disorders, so it is likely that reported disability levels, while high compared with non-care leavers, could still be understated.

Too few care leavers identified as coming from an Indigenous background or as being from a non-English Speaking Background for us to be able to draw meaningful conclusions regarding the representation of care leavers within these groups. Existing evidence highlights that Indigenous people are extremely over-represented in out-of-home care across Australia, and significantly underrepresented in higher education. How these patterns are reflected in the participation rates of Indigenous care leavers at university requires further data and research.

#### Achievement:

Our analysis shows that, once admitted, the academic performance of the care leaver sample was relatively strong when compared to non-care leavers. Using the Department of Education and Training's success rate measure, care leavers reported a pass rate of 93 per cent, compared to 89 per cent for non-care leavers. Based on the achievement measure of Weighted Average Mark, we found that care leavers reported an average mark of 71, compared to an average mark of 66 for non-care leavers. Taken together, these figures tentatively suggests that care leavers who access university perform relatively well. Success at subject level may reflect qualities such as resilience, perseverance and maturity, which our qualitative research suggests are held by many care leaver students at university. However, more data are required, particularly around retention and completion, to understand the longer term effects of external barriers on graduation rates beyond subject success.

#### Section summary

The collection and monitoring of data is central to developing policies that can attract and support care leavers in higher education. Given the ongoing absence of care leavers from the Australian higher education student equity framework (Harvey, Burnheim & Brett (eds) 2016), policies at institutional and state level are necessary for information gathering. The project team was able to advocate and influence change at multiple levels, including contributing to the establishment of specific questions on statewide educational access scheme forms, university enrolment forms, and university bursary application forms. By ensuring consistent language across each context, the team has provided a template for further expansion and highlighted the feasibility of multiple data collection techniques. The importance of identifying care leavers early in their enrolment is further revealed by the interviews with students (Section 3), where reluctance and lack of capacity to access services is a common theme, but where the benefits of such services are also extolled by those able to access them. The project has highlighted the capacity of institutions to capture care leaver data at multiple points of application and enrolment, and the potential for such methods to be expanded across the sector.

In addition to the establishment of measures that will provide an important baseline of evidence for the sector, specific data were collected on the enrolment patterns of care leavers from one partner institution. The data revealed a propensity of care leavers to enrol in disciplines such as Health Sciences and Education, the relatively high likelihood of care leavers to be female, regional and/or from low SES backgrounds, and the relatively high success rates of care leavers in subjects enrolled. This last finding in particular is notable, despite clear limitations of the sample size. Many care leavers are able to succeed in higher education precisely because of the qualities of persistence, resilience and maturity that were underlined within our interviews (Section 3). Having established the necessary templates for improved data collection across the sector, a challenge for Australian universities is to develop these templates to improve access and success of care leavers. Meeting this challenge would reflect not only a commitment to social justice but a desire to enrol a diverse and representative student cohort, necessary to ensure the highest quality of learning for all students.

# Section 3. The voice of care leavers: findings from in-depth interviews

This section of the report presents the major findings from our interviews with enrolled university care leaver students. The semi-structured, in-depth interviews were designed to capture views on higher education issues such as: application and enrolment processes; transition into higher education; and support services at university (see Appendix for the interview guide). Interviews were conducted across the four partner universities with a total of 35 care leaver students. Participants were recruited for interviews via a range of advertising and promotion methods across the different institutions. Interviews were digitally recorded, transcribed and analysed for content and themes. An interpretative phenomenological approach to the analysis was applied (Smith, Flowers, & Larkin, 2009).

Table 1 presents a summary of participant characteristics and care experiences. The majority of participants were female students aged below 36 years.

		Number	Proportion
Institution	University A	12	34%
	University B	13	37%
	University C	2	6%
	University D	8	23%
Gender	Male	6	17%
	Female	29	83%
Age	18-25	13	37%
	26-35	14	40%
	>36	8	23%
Aboriginal or Torres Strait Islander	Yes	5	14%
	No	30	86%

#### Table 1: Summary of participant characteristics (n=35)

#### Participant care experiences

Table 2 presents a summary of participant care experiences. Nearly one third of the participants had experienced multiple care placements.

		Number	Proportion
Years in care	1-5	16	46%
	6-10	8	23%
	11-18	6	17%
	Missing data	5	14%
Multiple out-of-home care placements	Yes	11	31%
	No	24	69%

Table 2: Summary of participant care experiences (n=35)

Table 3 presents a summary of responses relating to the transition to higher education. The table shows that 46 per cent of the care leavers had not transitioned to university directly from school. In the interviews, mature age students typically reported not having any assistance, or having very little assistance, applying for higher education. These students reported finding the application process complex and difficult. In contrast, care leavers entering university directly from secondary school, who were typically younger students, typically reported receiving more help during the application process at school and having less difficultly applying for university. Interestingly, only around one third of participants sought compensation through an educational access scheme application, and only around half applied for an equity scholarship. Given the extent of disadvantage outlined in the interviews, it is likely that the vast majority of interviewees would have qualified for some form of compensation had they applied, so this evidence confirms the need for greater awareness raising and transparency of financial bursaries and compensation.

		Number	Proportion
Transitioned straight from school to University	Yes	16	46%
	No	19	54%
Educational Access Scheme Applicant	Yes	11	31%
	No	19	54%
	Unsure	5	14%
Applied for equity scholarship	Yes	19	54%
	No	16	46%
Applied for other scholarships	Yes	8	23%
	No	27	77%
Assistance applying for university	Yes	13	37%
	No	22	63%

Table 3: Transition to higher education (n=35).

Table 4 summarises how many participants had experienced homelessness, mental health issues, and community support. A total of 37 per cent of the participants had experienced homelessness at

some point, and 51 per cent identified as having a mental health disorder, most commonly an anxiety disorder. Some participants reported traumatic experiences prior to being in care, or during their time in care, which overshadowed educational concerns. Significant challenges such as homelessness and multiple placements caused major interruptions to school education. During difficult times, 57 per cent of participants reported accessing community services. Most of the participants, however, reported finding community services unhelpful. The high extent of disadvantage and trauma identified further confirms that any suggestion of simply accommodating care leavers within the 'low SES' equity category is unrealistic.

		Number	Proportion
Homeless at some point	Yes	13	37%
	No	22	63%
Identified as having anxiety or other mental health disorder	Yes	18	51%
	No	17	49%
Community support accessed	Yes	20	57%
	No	15	43%

# Table 4: Experience of homelessness, mental health disorders, and community support access (n=35).

Participants were asked to suggest the types of support they thought would attract young people in care, and/or care leavers, into higher education. Table 5 shows the top five responses. The most common response was school outreach programs (mentioned by 26 per cent of participants). Also notable was the proportion of respondents (23 per cent) who argued the need for greater education of carers, social workers and case workers. The need for this education was highlighted in *Out of care, into university* (Harvey et al. 2015), where researchers found a culture of low expectations across and beyond the welfare sector, and identified a range of barriers to the education of carers and associated staff.

#### Table 5: What type of support would encourage care leavers into higher education? (n=35).

	Number	Proportion
Outreach to schools	9	26%
Financial support services and housing options	7	23%
Educate carers, social workers, case workers	7	23%
Communicate information about services available	7	20%
Tailored support services, scholarships, and bursaries	4	11%

Participants were also asked to suggest the types of support that could be provided to care leavers already enrolled in higher education. Table 6 shows the top five responses, with the most common responses being counselling and mentoring, more accessible information and services, and academic skills and careers guidance.

	Number	Proportion
Counselling and mentoring	9	26%
Accessible information and services	8	23%
Academic skills and careers guidance	8	23%
Financial planning support	7	20%
Temporary accommodation support	4	11%

Table 6: How can universities support care leavers who are already enrolled in higher education?(n=35)

Participants were asked further details about their preferred types of support that could be available to care leavers at university. In terms of financial support, the majority of participants preferred direct payment over vouchers. Participants indicated that direct payment would allow flexibility to meet their most urgent financial needs when required, rather than having to rely on a specific voucher. As primary concerns are often accommodation and financing the costs of living, direct financial aid is recommended as a mode of support for care leavers.

The possibility of an academic adviser program was popular among participants, with only one participant opposing the idea. The possibility of having a mentor who could advise on academic skills and curriculum appealed to participants. However, many were concerned about not getting along with their mentor as they had previously been in similar mentoring situations which had not worked out well. Many participants thought the mentoring program should not make participants feel like they were being monitored, should not be compulsory, and that different mentors should be made available if the initial pairing was not productive. Further research on the efficacy and flexibility of the various academic advising and mentoring schemes across Australian higher education could help to inform new models that are suited to increasingly diverse student cohorts such as care leavers.

The idea of establishing a peer group for care leaver students was met with mixed responses. While students identified that such a group could be useful, they also shared many concerns. It was generally thought that the group would need guidelines and parameters around the sharing of past experiences in care. There was a concern that participants would be expected or encouraged to disclose personal and private information in the group or that others might talk about traumatic incare experiences causing re-traumatisation to others. Responses were influenced by the type of experience participants had growing up and socialising with other people in care. A strong factor here is the complexity of the identity/stigma issue connected to being a care leaver; some people want to leave that fragment of their identity behind and 'just be a normal student'. Other participants have integrated their care experiences into their present self-narrative and are more comfortable identifying as care experienced. Broader social prejudice and misconceptions about being in care impacted on the way participants identified with their care experience.

Table 7: Preferences relating to financial support, academic advisors, and introducing a care leaver group (n=35).

		Number	Proportion
Finance support	Direct payment	26	74%
	Vouchers	8	23%
	Missing data	1	3%
Would like an academic adviser program?	Yes	34	97%
	No	1	3%
Would like a care leavers group?	Yes	20	57%
	No	9	26%
	With parameters	6	17%

## Thematic analysis of interview data

This section of the report presents a thematic analysis of the interview data. All interview data were de-identified and pseudonyms have been used.

#### Disruptive or traumatic experiences while in care

Traumatic experiences in care typically impact negatively on self-worth and self-confidence and socialisation. Many care leavers discussed their present challenges in higher education with reference to their experiences growing up facing a number of challenges. When asked about their pathway into university and what was going on in their life at that time, many decided to start from their childhood to explain the various events and choices that led to their enrolment in university. Counselling services had been available to some, but not all. The majority of participants had an understanding of the impact early childhood events had on their sense of self and the way they engaged with other people.

'I had always wanted to go to uni, and it had never been an option, kind of like an option that I couldn't achieve, but the situations that kind of arose close to that period obviously hindered me a lot ... the counsellor I was seeing said I had post-traumatic stress disorder again, like exacerbated from my childhood ... ('Angela').

Care leavers deal with the impact of traumatic experiences in different ways. Some are keen to face their issues and work through them, and others do not want to reflect on their past experiences. Many are keen to move forward and forget about the past. A major idea expressed by a couple of people was that 'your past does not shape your future'. Participants were asked about counselling services available at the university. Some people said they knew about the services but didn't think they should use them (not entitled, didn't want to burden the system, have had bad prior experiences with counsellors). Other participants said they did not think they needed to access these services.

Care leavers described their overall experience growing up in care as positive and supportive. However, support can also be paired with mixed messages and low expectations, as 'Zoe' highlights: 'I was really discouraged from doing it [university] because I was too far behind and I also didn't feel like it was something that ... was expected of me to do, like I don't know if that makes sense, but I kind of felt like it wasn't something my family had done and like it just wasn't how, what was expected of me, that low expectation. But I still wanted to do it and I think I had, I had a few foster carers particularly in year 10 and then in year 12, my foster parents were very encouraging and supportive of me of doing that, but a lot of it was me trying to actually believe I could do it' ('Zoe').

Support from the 'foster family', 'carer' and other individuals in accessing university is significant, as 'Jamie' describes:

'I think for a few years in my recent past I wasn't linked in with my foster family and that was a huge barrier as well. Like I don't know if I wouldn't say it would be unachievable for me to have gone to uni without their support, but it was like a million times harder and you know they helped me with printing out all my readings and all that stuff, and helped me with food when I was struggling. So if I didn't have their support still, I don't know if I would have even bothered trying' ('Jamie').

#### Financial barriers and accommodation

There are practical barriers which make entry to university difficult for care leavers. The most common barrier is the cost of living involved in undertaking university study. At the age of 18, when young people 'age out' of care, many will need to work to finance the costs of living and accommodation. At the age of 17, care leavers attending school are working towards their final mark which will potentially be used to determine entry to university. The stress and pressure that young people experience in this transitional phase is significant. In this context, the age of leaving care is one of the greatest barriers young people leaving care face in pursing educational opportunities.

'With working I was doing three or four night shifts a week, so I think that really contributed to why I failed first year as well, because I was just working, it was kind of just to make ends meet, because we had a fair bit of rent, you've got groceries, you've got bills, you've got electricity gas etc etc and then the lack of sleep for one, so then you don't really turn up to any of your lectures and tutes and then you fall behind in that area' ('Chris').

'The main thing would be money. During my undergraduate degree, I was working as many hours as I can... at one point I had to miss an exam because I got an eight hour shift at work, and I needed the money so badly - I just had to miss it' ('Asher').

Many of our interview participants worked to support themselves from an early age as they did not have a network to rely upon for financial support, even if other forms of support (such as accommodation or emotional support) were made available. This was true of care leavers across different age groups.

'Because the thing is that there gets to a point where something does happen, whether it's you need your wisdom teeth removed or you need this or you need that, like there will be something, you will get sick, you will have something happen and does the university really want people to abandon their studies, or turn around and go back to an abusive environment, because that is why every single person leaves, it's because of some form of abuse' ('Cara').

Many mature age students deferred applying to university, or decided they had to exit early, because they felt they simply could not afford it.

Centre for Higher Education Equity and Diversity Research latrobe.edu.au/cheedr

'I talk to some of my classmates that are still living in community housing and you know there was one young lady - she's 25, she's got 2 kids, she's been struggling and she's had sick children and that, and only the last couple of days she's told me that she's dropping out of uni which I'm a bit sad about. And I mean others don't understand it - they go home and dinner's cooked for them whereas some of us older ones have got to go home and cook dinner and go through all that sort of sh\*\* and still maintain the adult life' ('Matt').

'The expensive part about uni isn't the actual - it's more about where you're living rather than how much you spend on uni ... [and] at uni you spend money to print off stuff, you have to spend money on other things' ('Dale').

'I was travelling between Bendigo, that's a 6 hour round trip every day, so I was pretty much there for class and then straight gone' ('Alexis').

'I think transport, I found it an issue, but that was only because I lived on the north side of Brisbane and I was going to uni on the south side of Brisbane, so that was one of the reasons why I had the motorbike' ('Sophie').

#### Lack of formal assistance and information when applying for university

One of the biggest challenges for those who are not guided through the application process is not knowing exactly how to apply. There was also a lack of knowledge about the application process among carers. 'Dale' says he would have better prepared for the application process if he had been involved in the process while still in care.

'The messing around with all the paperwork is the hardest part because we don't do any of that, because we're just told to go this way that way when we're in foster care, whereas they should be able to make their choices on what's going on and stuff' ('Dale').

Participants compared their experience of applying to other students and said that a major factor was not being able to rely on family to guide them through the process because many carers had not been to university. In this context, schools played a pivotal role in providing information about higher education. Higher education was made most accessible by schools discussing the application process in class and even making it a classroom activity. Careers counsellors were somewhat useful for communicating information but not across the board. Some schools were described as 'good' and 'supportive' by participants because they took an interest in student wellbeing, future planning for university, and academic performance. Other schools were described as 'not very good', uninterested in student performance, and doing little to encourage future planning.

'Other people, they might have family support and there's other times where they might not have that family support but they've still got family if they need money or need help their parents can do it, whereas I had to try and find a place to stay ... I had to buy all my text books with my money from Centrelink and that was like \$500 so that put me down a lot really quickly and so I had to sign up a lot for scholarships and it made it a lot harder during the very early transitioning days' ('Dale').

From year 10 to year 12, many young people had concerns and responsibilities which overshadowed university, or they did not think they would be able to get into university. They could not imagine university as a viable or possible future. Many participants recalled applying for university at the time of leaving care, often without formal assistance doing so. The lack of formal assistance applying was more of a problem for mature age students, and there were issues here about the accessibility

of information more generally. Those students transitioning straight from school found the process straightforward if they had received assistance, but not all schools are helping their students apply.

'My HSC I was an emotional wreck, I was vomiting every day and I just wasn't coping with the situations at all' ('Angela').

'There's so much that you're thinking about, well that I was thinking about, when I was leaving care you know because my birthday's in the middle of the year as well and DHS were like, "your payment gets cut off in May", and you've still got the rest of the year to go so I was like, I wasn't going yay I'm going to go to university next year that's gonna be great fun. You know it was kind of like it was one of the last things that I was thinking about, and I was only thinking about it because I had the foster parents who were encouraging that' ('Kelly').

A small number of care leavers in the younger age group were guided through the application process by their schools and found it straightforward to enter university from high school.

'I think it was online and our school did a lot of it. We were very supported in the way that we didn't have to do things - like we got given our numbers and all that kind of stuff, we didn't have to search for things. I don't know if it's common that the schools do it all for you but they did, they enrolled us and they clarified that this was right - it wasn't like, you know, can you fill out all this information all over again. If they had the information they just sort of processed it for us' ('Brooke').

Those students transitioning from school found the process straightforward if they were helped, but not all schools were perceived as helping their students to apply.

'My partner went to uni, and he helped me through a lot of that part. But if I didn't have him, I wouldn't know how to enrol or do anything. I'm not very technologically savvy. I would have been lost without him' ('Chloe').

'Even like basic things, so there's a lot of communication through websites and having to pay for parking permits all that sort of stuff I didn't know, so I had to learn all on my own like I was the first in my family to go to university so it was all really new to me' ('Brooke').

During the application process, some participants in the older age groups were concerned about the academic skills required. This concern meant that some participants put off applying to university and other experienced feelings of nervousness.

'I guess having had my last experience of education so long ago, and I was so disengaged from my education when I was there, it was a little bit nerve wracking but there weren't any barriers or difficulties' ('Samantha').

Mature age students who did not enter university straight from high school discussed their pathways into university. For some participants it was a matter of always wanting to go to university but having to wait for the right conditions, personally and in terms of financial circumstances, before seeking out information about the application process.

'So then I essentially enrolled in night school to try and improve my grades ... during that time I heard about industrial design, and I thought yeah that's the ticket for me. So when I finished, I did chemistry and English, I was about to do physics, but I heard that it wasn't necessary to do industrial design, so I applied to do industrial design, my grades weren't terribly good, but because I was older than 25 I could get in through alternate entry, and I did' ('Kurt').

The majority of participants did not enter higher education directly after finishing high school. Alternative entry points to higher education, such as bridging and enabling programs and articulation from TAFE courses, were popular among mature age students and those students not applying directly from school.

'The Tertiary Enabling Program, it's amazing. It's a 6 month bridging course available to anybody. I think it's more than a couple of years outside of high school and it's a 6 month, just one semester, course made up of four subjects there's a foundation in maths, a foundation in science, an academic integrity module and a critical thinking module and getting a certain score in that will give you an automatic pathway into university' ('Samantha').

'I came back to Victoria 5 years ago, and I'd already started TAFE over in Western Australia and did Cert III and IV over there, so I came to Bendigo and I enrolled in my diploma at TAFE. So I did my diploma of community services and then I came through the TAFE system through to Uni into my second year' ('Dianne').

#### Academic skills and experience

Many care leavers revealed a lack of academic preparation upon entering their undergraduate degree. Some of the skills highlighted, such as essay writing and referencing, are explicitly taught within university enabling programs, and it may be that some students would be better directed to those programs as pathways, where available, rather than entering undergraduate degrees directly. Nevertheless, our quantitative data and further interview responses suggests that many care leavers are able to work through their academic skills issues and successfully adjust to university study. The comments below highlight the initial struggles faced by commencing care leavers:

'I think my first year I really struggled with the academics because that's the part that to me the university has been the hardest, is just, because I don't have a lot of my basic skills so I've kind of had to play catch up, and that means that sometimes [I] particularly have a hard time putting words down onto paper and sometimes it's just daunting' ('Zoe').

'I had to teach myself how to research and write essays and so on. They provided me with a lot of guidance, because I didn't even know how to use books or the internet properly' (Peter).

#### The impact of low expectations on self-worth and self-confidence of care leavers

*Out of care, into university* (Harvey et al. 2015) highlighted a damaging culture of low expectations that often surrounds children in care, including from carers, case workers, and teachers. At least two of our interviewees were told by school teachers that they should not aim for university. Alexis tells a particularly disturbing story which reveals the lack of support young people receive when their problems become 'too big'. 'Alexis' describes living in an abusive home and making this known to her teachers. The teachers considered the challenges 'Alexis' would face living out of home and participating in higher education and decided it would not be possible. 'Alexis' was told that university would probably not be possible for her unless she stayed with her parents. She attempted to stay at home while she completed high school but found herself studying on the streets because the situation at home was so bad. 'Casey' describes the impact of her teacher having low expectations on her educational experience:

'In high school, we did have a careers advisor but he literally said to me "you will never get into medicine, pick something else". Instead of saying well here are the pathways to medicine. It was you'll never do it ... every time I came out of the careers advisor I was always in tears, I always waited 'til I left the room to cry, because I wasn't giving him the satisfaction of upsetting me ... I should've been a cook or like a cleaner in his opinion' ('Casey').

Despite these challenges, university was seen as way to turn your life around, to build career prospects and a future. For some participants, the achievement of being at university (and doing well) was enhanced by the fact that they had been told by high school teachers that they probably wouldn't get into university.

'My sister and I talk all the time - like my sister just got offered a head nurse position, and she's only in her third year out of uni, and she's like "I just love seeing people in the street, even past teachers and stuff". Everyone that ever said you won't get anywhere or you won't do anything or even just looked at you and we say the same thing to each other all the time you know how nice is it to just turn around and rub it in everyone's faces that said that we can't do it and we're doing it' ('Jane').

The low expectations that care leavers often face are very closely tied to individual narratives of accessing university and building up enough confidence to apply in the first place. As 'Jane' says, showing people she had known in the past just what she was capable of was quite a rewarding experience. These are the people who underestimated her. 'I'm actually quite smart', she said in a matter of fact way. However, confidence and self-worth remained issues in the background with which many care leavers were dealing:

'overall like the self-esteem in its true sense wasn't there, but I also had additional pressures that other students who were living at home with their mum or their dad or their grandparents even didn't have ... like to get a car I had to save myself, you know even though I was living with my friends and I now call mum and dad, I still had to prove, I felt like I always had to prove something, prove that I was worthy of being there, prove that I was worthy of them helping me' ('Angela').

'... I had a lot of self-worth issues, I didn't think that I was good enough ... I think that prevented me from really pushing myself to try and get into medicine, I gave up pretty quickly, and there are other pathways I could've taken' ('Casey').

Building self-worth and confidence may have a significant impact on the ways in which care leavers imagine their potential. Those care leavers who thought that they would not be accepted to university, and/or that they should not apply, were able to build confidence throughout their time at university by learning, being supported and encouraged, and discovering through experience that they could be academically successful. For some participants this did not come as a complete shock because they achieved high grades in high school, though still were unsure of their ability to participate in higher education. Many spoke about turning their lives around and being able to participate in higher education despite the obstacles or disincentives they encountered growing up, such as being told they would not be able to get into university.

Diminished self-worth was a barrier to taking up opportunities such as applying for scholarships or utilising university support services. The responses below help to highlight why only around half of care leaver university students identified in our quantitative study had applied for an equity scholarship, and only one third had applied for compensation through an educational access scheme:

'I've always just thought scholarships you'd have to fill out hundreds and hundreds of pieces of paper and probably get rejected so I've never bothered' ('Jenny').

'There's always someone more worse off who deserved it more than I did, so I sort of would leave it' ('Grace').

'I didn't go for any scholarships because I didn't feel like I was worth them and there were significant barriers because I didn't actually speak up. I didn't say "look I need this or that" because I was 17 and I wasn't raised to be particularly forward in what I needed and there was also an element of shame there was a very strong element of shame so it wasn't something I spoke about' ('Helena').

These responses reveal the need for secondary schools, tertiary admissions centres, university scholarship offices, and support services to improve awareness of bursaries, compensation and eligibility requirements, and to improve the transparency of processes and ease of application. Clearly, many prospective and enrolled university students who are entitled to significant compensation are not applying or receiving that compensation, and this gap affects university access, success and outcomes.

Interviewees also provided advice for prospective students, which was often oriented around selfworth and competency and turning your life around.

'Where you have been does not define who you are, there is no you, you are who you are in the moment and you can choose who you are in the moment. Yeah you might be a bit behind from other families, but everybody's got their struggles and you are no less important and worthwhile than any other person, and if you want to do something just do it' ('Casey').

#### Resilience and strengths

Care leavers who had managed to overcome the barriers to university access were often strengthened by their experience and resilience. Many felt that their life experiences provided them with greater insight into complex social issues, insight their peers did not necessarily share.

'I think it's more life experience despite being the same age like you've done more, the rougher your life has been you get a little bit more mature and you get more understanding of how the world works' ('Dale').

Participants also spoke about their own resilience. As 'Trish' put it:

'Resilience ... Resilience I think is a big one because you don't sweat the small stuff when you've come from this kind of background. Little things don't matter to you' ('Trish').

Many care leavers balance university with paid work and family responsibilities. This means often being time-poor and exhausted, but also demonstrates a strong work ethic.

There were a broad range of motivating factors and past experiences that propelled students to fight for their future. These factors included advice from mentors like other family members, being responsible for other family members and children, the desire to do things differently and have a better life in comparison to early childhood experiences, indignation at being told they could not do something by a teacher, or being made to feel they were less than worthy. Certain experiences, such as raising children, also enabled participants to experience a sense of autonomy, productivity, accomplishment and self-worth. Participants explained building life skills and resilience in difficult circumstances like balancing competing commitments while studying. For many care leavers, this was a point of difference between their everyday experiences at university compared with other students. 'Chloe' explains:

'A lot of my friends who are at Uni don't have to work or worry about anything. Mostly because they have back up plans – they have parents if they need to rely on them. I just have myself. So in terms of paying all my bills, and having a house, and managing my finances I am solely responsible. I don't have the option not to work while I study' ('Chloe').

These differences in everyday university/life experiences sometimes mean that care leavers feel alienated from the 'student experience'. This speaks to the way in which student life is imagined and communicated/represented to students at university through, for instance, advertising campaigns and on-campus activities and groups. Care leavers and their experiences are typically not represented. Many care leavers would like to be treated 'the same', not despite their care experience but in recognition of it. This requires greater on-campus awareness and understanding of what being in care means. Through this type of knowledge stronger peer to peer relationships may also be generated.

'My two parents, that I did live with prior to being in foster care, they did put a high emphasis on our education and they really motivated us to want to sort of do well and I guess having been through that struggle during foster care, like my foster carer didn't have a job, so she was a full time carer I guess you could say. And we often struggled - like we lived in a poor living environment and that kind of thing and I think to myself in my own life I don't want to have to go through the struggle, I want to be able to be successful based on what I can do personally' ('Brooke').

'I think having kids was the biggest shift, I didn't believe I was – I always thought I'd be a good mum, but I guess having kids has made me realise there's so much decision making, and I actually can make those decisions for another person and myself I guess, and I think that's probably been the biggest thing that's helped me, having kids, because it's shown me that I can do that' ('Casey').

Such responses confirm that university awareness-raising and support programs need to avoid approaches of 'support with a measure of pity', as this is not the kind of support care leavers appreciate. Many care leavers found that in group conversations in class, with peers, and with administrative staff, that they were pitied. This experience of pity and condescension signals a deficit attitude/approach to care leavers. The idea that they have fewer resources and are therefore not as resourceful as other students is contradicted by the resilience care leavers demonstrate as they overcome significant barriers to participate in higher education:

'I have a lot more freedom and ability to study, and work hard, I feel I can pretty much adapt to most situations. So you know, knowing that time put in equals results, you know doesn't really matter what it is, that kind of can really be applicable to a lot of different things. So I think that is quite important, I think that has, I have that advantage over other students, in that I can just like you know, I can sit down and spend 2 or 3 days 12 hours, not 12 sorry, 10 or 11 hours a day just smashing out an assignment or trying to understand concepts ... I do have a lot more freedom to be able to leave when I want, come home when I want, I don't really have anything holding me back' ('Chris').

#### Peer and institutional knowledge about being a care leaver

Some participants noted that other students sometimes complain about seemingly trivial issues and this makes it hard to build relationships and establish common viewpoints with others. Relatedly, there can be a lack of knowledge and understanding about what being a care leaver entails. A lack of institutional knowledge inhibits the level of targeted support care leavers receive on campus and a lack of general knowledge about care among the student body can be a barrier to the development of friendships on campus for care leavers.

The lack of understanding from others, combined with the hesitancy to speak about out-of-home care, has an alienating effect for many care leavers. Some students found it difficult to share their experience in class, which can be a class discussion activity in social work courses. In this context the responses and reactions of other students can be problematic. Many interviewees felt that the attitudes/opinions expressed by other students in class were naïve and revealed underlying discriminatory attitudes.

'I think it probably contributed to an overwhelming frustration with the idealistic young people that I was studying with ...they made certain assumptions about people in foster care ... and I think probably struggling with the idea that I was probably at uni because of a really gross administrative error, and that I'm actually here, because I'm not one of them' ('Alexis').

Being in care was seen to be something that cannot be easily shared, is personal and private information, and was something that many perceived made them different from other 'normal' students. It was thought that normalcy is created (institutionally and socially supported) around many other student experiences and groups, but being a care leaver does not seem to be one of these categories.

'Honestly I think the most difficult thing for me is probably the whole social part to it. Like I have no problem with talking to people in class but it's the whole making friends concept, the whole socialising like asking people "hey do you want to hang out after this?" or whatever. It's kind of like I'm not an introverted person, I'm very out there, but I just feel like I have been affected in a way that prevents me from socialising in a way that I want to' ('Olivia').

'It would be easier to use cash - I'd be embarrassed using a voucher like it's some kind of hand out but if I'm with all my friends and we're buying materials at the book shop and they use cash and then I'm like oh I'm just using my special foster care voucher that would be mortifying' ('Amy').

'I think the biggest point is just making sure it doesn't feel like pity or you don't feel like you're being singled out because you're different or you're not good enough and you need to be propped up' ('Amy').

'I just want to try and be normal like everyone else kind of thing' ('Kelly').

'I left home at 14, I didn't have the skills, and lecturers and tutors were one of the first offenders in terms of making you feel like sh\*\* because they just assumed because everyone lived at home, everyone had come from school and had not experienced care, had not experienced abuse. So there was an expectation that you just went to school and then you got into uni and everything was hunky dory' ('Helena').

Many care leavers are enrolled in nursing, social work, psychology or sports health. Some of the content in these classes can trigger emotional responses from students in class. Triggers are often

related to personal experiences and can come about when a student personalises the learning material. Some participants experienced emotional, physical, or sexual abuse and violence in their childhood and have had strong reactions to course material in psychology and social work.

One student was removed from her psychology course and referred to a mental rehabilitation institution. Her experiences in care were very traumatic:

'That's when I changed over and thought I'd do counselling instead. And then that's where it all started, it all triggered, I started psychology, and that triggered all my childhood memories of all the abuse and sexual abuse, the psychological, the emotional, the physical abuse, it triggered it all... But when I was doing the human service degree it was not triggering, it was more like I was being a facilitator with all the skills that I already had and I was just helping other people, so I was doing what I enjoyed doing you know like being an interpreter with sign language, so it was more sort of conducive at the time, it was a form of healing.' ('Sophie').

'My social work degree and I guess they do touch on a lot of child protection matters and I guess sometimes it's a bit – it hits home a bit hard ... I have seen a lot of kind of stuff that kids have gone through so I guess that's it' ('Hannah').

Support on campus can play a major role. Care leavers were aware of counselling services, but the usefulness of counselling services received mixed responses. For some people, the option of having a counsellor on campus was helpful. Counselling was considered one of the best ways to get through time of stress or deal with issues that seemed overwhelming.

Care leavers said it would be helpful to have a liaison person, such as a care coordinator. The value of the care coordinator is that they 'get it', they understand what being in care means and can direct care leavers to relevant resources. Many care leavers have experienced the burden of paper work from an early age and are fatigued with this process, hence the importance of a contact person who can advocate on behalf of the student.

#### Support for Aboriginal and Torres Strait Islanders

Aboriginal and Torres Strait Islander care leaver students found Indigenous centres to be an important source of support. These centres often provide free practical support, such as tutors and internet access, as well as providing psychological support and a 'comfort zone' for students. Indigenous centres are also able to direct care leavers to related campus services, providing a single point of contact that was frequently valued. Given the over-representation of Indigenous children in out-of-home care, it is crucial that university outreach programs, and subsequent campus support programs, are developed with the guidance of Indigenous community members and in culturally sensitive ways.

'Everyone gets an equal chance and even me with a foster family. With my Aboriginality as well I actually think I get more resources as well because I get free tutors like I can organise keystones to success. And if I want to go in and I don't understand something I can have someone to talk to and we sit there for an hour and work that out whereas other students don't get that' ('Dale').

'...With the [Indigenous] unit, with the Aboriginal students there, at least I can go there and I've got somebody who I can bounce off my knowledge and feel a bit, you know, just feel a bit more comfortable and not giving up and not quitting, because there is that support mechanism there...' ('Sophie')

## Section summary

The care leaver interviews revealed several major barriers to accessing higher education including high rates of homelessness, financial problems, and mental health issues, many stemming from periods of childhood. Despite the existence of these barriers, interviewees were relatively reluctant to apply for university compensation or bursaries, and mature age care leavers in particular also experienced a lack of support with university application processes. Participants recommended a range of approaches for increasing access to higher education for care leavers, including expanded school outreach programs, financial support services, housing options, and greater education of carers, case workers, and social workers. The need for a culture of heightened expectations among some school teachers, carers and related staff was also highlighted.

During higher education, concerns raised included accommodation difficulty, high costs of living, perceived academic skills deficiencies, and low self-confidence. Many care leavers experienced issues with self-confidence, and at times felt misunderstood by other students, and experienced feelings of alienation. Issues raised suggest further attention is required to improve campus climate, culture, and curriculum particularly in accommodating and understanding diversity. Care leavers, however, typically displayed high levels of resilience and emotional maturity compared to their peers. The interviewees recommended approaches for better supporting care leavers in higher education, including counselling and mentoring, more accessible information and services, and academic skills and careers guidance. For Indigenous care leavers in particular, presence of a single point of contact on campus (usually through a dedicated Indigenous centre) was seen as valuable.

# Section 4. Higher education guidelines for professional and academic staff working with care leavers

The following guidelines are designed for people working with care leavers in the Australian higher education sector. Care leavers are people who spent time in out-of-home care, for example foster care, kinship care, or residential care. In this guide you will find information about the different forms of support that professional and academic staff can offer care leavers. The guidelines provide advice for university staff delivering outreach and in-reach programs to young people in out-of-home care, and outline the support services available to care leavers who are studying at university.

The guidelines are informed by research conducted by La Trobe University, Federation University of Australia, Western Sydney University, and Queensland University of Technology, for the *Recruiting and supporting care leavers in Australian higher education project* (Harvey et. al, 2017), The research aimed to raise the educational aspirations of young people in care, and their support networks, and develop support services for care leavers in higher education. As part of this project, 35 care leaver students from four different universities in Australia were interviewed, and their firsthand accounts have helped to shape these guidelines. Please note all interview data were de-identified and pseudonyms have been used throughout these guidelines.

While written for staff in higher education working with care leavers in Australia, the guidelines may be relevant to other stakeholders and contexts. While this is not a new area of research within Australia, the provision of specific services for care leavers within higher education is inchoate. The guidelines could provide a basis for the development of further resources for supporting care leavers in higher education.

The guidelines are designed to be read alongside '*Higher education guidelines for students who spent time in out-of-home care*' which was written for care leavers (see Section 5 of this report).

Disclaimer

The information contained in these guidelines is of a generic nature only. You must not rely on the information in these guidelines as an alternative to seeking advice from an appropriately qualified professional and/or making your own enquiries to other higher education institutions to confirm the accuracy of the information provided. The information contained in this publication is current as at the time of the publication (March 2017). The University reserves the right to make changes without notice, at any time in its absolute discretion, including but not limited to varying these guidelines. *These guidelines must not be published, quoted or disseminated to any other party without the University's prior written consent*. To the extent permitted by law, the University does not accept responsibility for any loss or damage occasioned by use of any of the information contained in this publication.

- These guidelines are designed to provide professional and academic staff with information to help support care leavers in higher education.
- After leaving formal care, young people are often referred to as 'care leavers'. Not all care leavers identify with this term, however, and may think of their experience in terms of having been in 'out-of-home care', 'foster care', 'kinship care' or being 'care experienced' or 'looked after' (NNECL, 2013, p.11-12).
- Formal support for young people in out-of-home care decreases at age 18. Care leavers are often faced with uncertainty around where they will live and how they will manage their financial independence.
- Some care leavers will have access to support networks and formal pathway plans outlining educational and career goals, while many others experience the 'soft bigotry of low expectations' which impacts heavily on their educational aspirations (Harvey et al., 2015b:, p. 6).
- University outreach and in-reach programs for care leavers are best developed in consultation with non-government organisations and agencies in the out-of-home care sector.
- It is important for services and divisions of the university to work together to provide 'wrap-around' support, which requires strong links between services, and 'embedded' support, where support for care leavers is part of broader, sustainable practices.
- To support care leavers, professional staff can facilitate valuable links between Equity Units and areas such as Admissions, Marketing and Recruitment, Student Wellbeing, Accommodation Services, Scholarship and Advancement, Alumni and Advancement, Counselling and Careers Services.
- Some care leavers can have mental health issues, especially anxiety, tied to their past experiences. It can be helpful for these students to have access to counselling and mentoring, and flexibility from academic staff in regard to assessment.
- Many care leavers have backgrounds and life experiences which differ substantially from students who have not lived in care. Care leavers can therefore provide valuable and unique contributions to classroom discussions, though sensitivity and training may be required from academic staff.

#### Student voice:

'I was really discouraged from doing university because I was too far behind and I also didn't feel like it was something that ... was expected of me, I don't know if that makes sense, but I kind of felt like it wasn't something my family had done and it just wasn't what was expected of me, that low expectation'.

'But I still wanted to do it and I think I had, I had a few foster carers particularly in year 10 and then in year 12, my foster parents were very encouraging and supportive of me of doing that, but a lot of it was me trying to actually believe I could do it'. ('Zoe').

#### **Background information**

Out-of-home care is a term used mostly by professionals working in the community and government sectors to refer to the care of children who are unable to live with their families (Peel & Beckley, 2015, p. 372). Out-of-home care involves the placement of a child with alternate caregivers on a short-term or long-term basis. In Australia, formal out-of-home care ends at 18, with some young people as young as 16 leaving care (Victorian Department of Human Services, 2007; Mendes, 2009, p. 34). After leaving care young people are often referred to as 'care leavers', however not all care leavers identify with this term and may understand their experience in terms of having been in 'out-of-home care', 'foster care', 'kinship care' or being 'care experienced' or 'looked after' (NNECL, 2013, p. 11-12).

Knowing how to provide suitable support in a higher education environment is crucial to the success and well-being of care leavers. Care leavers often face significant financial responsibilities which they balance with study and social commitments. Many care leavers are self-reliant and skilled in balancing competing demands on their time. Given the high demands on the time of care leavers, however, having readily accessible information and support is paramount. These guidelines are designed to provide professional and academic staff with information to help support care leavers in higher education.

Within Australia there are approximately 43,400 children in out-of-home care and this number has risen steadily over the past decade (Australian Institute of Health and Welfare, AIHW, 2016). Young people leaving care are one of the most vulnerable groups in our community (AIHW, 2016). Care leavers typically experience severe educational disadvantage and are disproportionately likely to come from low socio-economic status, regional, and Indigenous backgrounds (Harvey et al., 2015a; 2015b).

Ageing out, or leaving care, occurs at a time when many young people might otherwise be looking at educational options. Instead they are often faced with many decisions such as where they will live, how they will manage their financial independence, find employment, manage relationships, and seek other forms of appropriate support (Peel and Beckley, 2015, p. 376). One concern is that, during this period of transition to independence, care leavers often miss out on higher education and training opportunities which would lead to greater employment prospects as well as increased earning potential (Lomax-Smith, Watson, & Webster, 2011; Norton, 2012).

Some care leavers have access to support networks and formal pathway plans outlining educational and career goals, while many others will experience the 'soft bigotry of low expectations' which impacts heavily on their educational aspirations (Harvey et al., 2015b, p. 6). Support and peer networks can offer care leavers a significant resource which should not be overlooked, but development of specific university support services is also required to improve the accessibility of higher education for this group.

#### Outreach and in-reach programs

University outreach and in-reach programs are typically aimed at students in primary and secondary schools and designed to: familiarise these students with university life; raise awareness of, and aspirations for, higher education study; and/or inspire interest in particular disciplines. Outreach involves staff from the university delivering advice and services off campus, often within the school grounds. In-reach involves bringing students onto university premises. Universities can specifically target outreach and in-reach programs to students in out-of-home care, for example by working with community services organisations and flexible learning schools.

For care leavers, university outreach and in-reach programs are best developed in consultation with non-government organisations and agencies in the out-of-home care sector. <u>The Reach Program</u>, for example, was developed by Western Sydney University's Widening Participation Unit and Life Without Barriers to encourage young people aged between 15 and 17 years to consider university or

other tertiary education. Participants completed a three month program that included: a range of activities to gain exposure to university life; resources to support students if they chose to go to university; and career coaching. The program also involved the engagement of foster carers and collaboration with other not-for-profit organisations, Dress for Success and the CREATE Foundation.

When delivering outreach or in-reach, it is helpful to consider these major components: location; the message; audience; event structure and teamwork; and content. The following table describes why these components are important in the context of outreach and in-reach to care experienced young people.

Location	<ul> <li>In-reach can be delivered on-campus, but some students may be more comfortable in their own schools or on common ground.</li> <li>Outreach may be an effective way of familiarising students with university life with minimal inconvenience to prospective students, carers and other staff.</li> <li>Logistics can be complex given the involvement of multiple stakeholders, e.g. agency staff, carers, teachers, and students.</li> </ul>
The Message	• Outreach can raise the expectations of both care leavers and their support networks, transforming the question 'Can I go to university?' into 'Why shouldn't I go to university?'
Audience	<ul> <li>It is important to ensure that care leavers and their support networks are informed about what university involves and what support is available.</li> <li>Support networks may include carers, friends, case managers, counsellors, grandparents, and other family members.</li> <li>Information and presentations should always be age-appropriate.</li> </ul>
Structure and Teamwork	<ul> <li>People participating in outreach may appreciate having their support networks and peers present at events, and being teamed up in group activities with friends.</li> <li>Outreach events can be organised to offer concurrent information sessions and activities for care leavers and their support networks.</li> </ul>
Content	<ul> <li>Outreach presentations should provide information on the types of support available to care leavers at university. Financial and accommodation support are often primary concerns.</li> <li>Be clear and specific about what can be involved in attaining a degree, including study time and on-campus commitments.</li> <li>Discuss the benefits of a university education, e.g. university graduates are likely to earn a higher income than non-graduates.</li> <li>Including care experienced people in outreach programs, for instance as presenters, can provide a reference point for people who have grown up in difficult circumstances.</li> <li>Care experienced presenters may not wish to share various elements of their personal story so it is a good idea to establish parameters around question time.</li> </ul>

#### Good Practice Case Study: Outreach at the University of Sheffield

The University of Sheffield offers Outreach through its 'Looked After Children & Care Leavers <u>Project</u>' (2017). The program provides aspiration raising activities for young care experienced people. These activities include:

- 22-week 'Building Learning Power Club' Y5/6
- 3-day summer school Y10/11
- 'Routes into Further Education' Y11
- 'Discover University' day Y12.

Information about the program, including how to apply, is communicated to young people in care and their carers through community and government organisations.

#### LAC Champions

The LAC Champions outreach program is supported by care experienced students attending the University of Sheffield. These students are known as 'LAC Champions' after the 'Looked After Children' Project (<u>LAC Champions' page</u>). These students convey a powerful message during outreach events: 'if I can do it, then so can you'. LAC Champions are provided with training and support so that they are able to:

- Work as mentors and work with student groups
- Assist in the delivery of the Y5/6 'Building Learning Power Club'
- Support summer schools and other outreach activities at the university.

'The range of activities on offer presents an opportunity for LAC Champions to develop key transferable skills such as teamwork, time-management and communication'.

The University of Sheffield was awarded the Buttle UK Quality Mark for the support provided to care leavers (<u>Students from Care' page</u>). This support includes:

- A dedicated care leaver contact who can offer tailored, individual and sustained support
- Help in accessing financial help, guidance and support
- A generous Care Leaver's non-repayable bursary
- Priority in obtaining University owned accommodation.

#### Wrap-around and embedded support services

It is important for services and divisions of the university to work together to provide 'wrap-around' support, which requires strong links between services, and 'embedded' support, where support for care leavers is part of broader, sustainable practices. Developing support services for care leavers requires understanding *where* support can be most effectively targeted, and it is best to include care leavers in consultations about their support needs and preferences. Effective support is sustainable and creates a reliable network for care leavers.

The combination of wrap-around and embedded support can:

- Provide sustainable support services for care leavers
- Raise awareness about care leavers among staff
- Raise the profile of care leavers within the student body
- Create a safe environment for care leavers on campus and a sense of belonging
- Encourage care leavers to utilise support services.

#### **Student voice**

Identifying as a care leaver can be difficult. During university, care experienced people of all ages may consider identifying as a care leaver for a variety of reasons, but also desire to have a 'normal' university experience and identify as a 'normal' student.

'I've grown from my experiences, so let's not sit there and have a pity party about it' ('Angela').

'I don't like people pitying me like I am in a situation – I think it's a good thing that what's happened to me, you know – I am who I am today because of it and I don't want people going "oh you poor thing" like that's not what I am about' ('Hannah').

#### Good Practice Case Study: The University of Michigan Blavin Scholars Program

The University of Michigan (U-M) has a dedicated care leaver program called the '<u>Blavin</u> <u>Scholarship Program</u>' (2017). The program provides assistance to students who have experienced time in foster care and who are undertaking a Bachelor degree at the University of Michigan.

The program draws on the <u>Fostering Success Coaching Model</u> which prioritizes partnerships between coaches and students. Each student is paired with a trained <u>Campus Coach</u> (a Blavin Program Staff member) who is also their primary point of contact in the program. Research shows that coaching and mentoring from older peers with foster care experience can promote acceptance and academic confidence (Phillip et al., 2015).

Campus Coaching comprises:

**A holistic approach** that uses the seven Life Domains (adapted from Casey Family Programs, 2006) to view Education and Academics at the centre of the Blavin Scholar's U-M experience. Using this holistic approach they acknowledge the impact the other life domains: Career Development, Financial Awareness, Physical and Mental Health, U-M Community Experience, Personal Identify Formation, and Life Skills Preparation have on Education and Academics and strive to promote skill development in each life domain.



Centre for Higher Education Equity and Diversity Research latrobe.edu.au/cheedr

#### The Blavin Scholars Program recommends:

- Regular coaching interactions and building a relationship of 'interdependence' between students and their supports.
- Empowering students by positioning them as the expert of their needs and experiences.
- Blavin Scholars 'set the agenda' by voicing questions and concerns with their Campus Coach.
- Students look towards the future, especially during coaching interactions.
- Developing academic, professional, and life skills.
- Supporting students in securing basic needs and resources to be successful.
- Monitoring academic and personal progress at university.

In addition to the support program, the **Blavin Scholarship** provides students with financial support toward the cost of attendance.

The program is a good example of how support for care leavers can be provided through a targeted program.

Applications for the program are taken by the university Undergraduate Admissions Office.

#### Want to know more?

There are many other targeted programs for care leavers including:

- The <u>Berkeley Hope Scholars Program</u> which provides support for care leavers enrolled at the University of California, Berkeley.
- The <u>University College London</u> which provides guaranteed accommodation for care leavers for 365 days per year, a dedicated member of staff for support/mentoring, and a bursary.
- The <u>University of Liverpool</u> which has a similar program for care leaver support with a bursary and a Care Leaver Coordinator.
- The <u>University of Bath</u> which provides guaranteed accommodation for care leavers for 365 days per year, a bursary and scholarship, and a Care Leaver Coordinator.

#### Self-Identification

Does your higher education provider offer care leavers an option to self-identify before or upon enrolment?

Some universities have methods that allow students to disclose their care leaver status, for example through application or enrolment processes. These universities can pro-actively contact their care leaver students with information about the range of support that is available to them.

Introducing a method of identifying care leavers through application and enrolment processes requires the cooperation and approval of staff across many different university divisions, including Senior Management, Admissions, Information Technology, and Student Information Systems.

Care leavers can be encouraged to disclose their care leaver status to access targeted support services. In the United Kingdom, for example, the <u>Propel</u> website was created to encourage students to inform higher education providers of their care leaver status in order to access support services (Who Cares? Trust, 2016).

#### **Contact Person**

Some universities have a designated Care Coordinator who: provides a single point of contact for care leaver students; provides general information and advice; and provides referrals to other support services. Some universities have staff within dedicated Indigenous centres who provide the point of contact for their Indigenous care leaver students.

#### Language

The following are examples of some terms that may indicate you are speaking with a person who is currently in out-of-home care, or has spent time in out-of-home care before they were 18.

- Care experienced Foster parent Residential care Kinship care Grandparent (or other relative) Ward of the State Independent living
- Lead tenant Carer Case manager Child protection Care leaver Looked after Aging out

#### **Accommodation Support**

Financially subsidised accommodation services provide one of the greatest forms of support for care leavers, particularly for those who are young and have not yet established safe and secure accommodation. Because care leavers often finance their own accommodation and living costs, this form of support can make a big difference.

On-campus accommodation services can potentially reserve a set number of rooms for care leavers for each enrolment period. In addition, university scholarship and advancement teams may be able to secure funding for accommodation scholarships, which may involve funding from philanthropic organisations or individual donations.

#### **Financial Support**

While many students are supported by income-contingent loans (e.g. HECS-HELP) and Austudy schemes, additional financial support can be crucial for living costs, the purchase of educational equipment such as books and laptops, and participation in social activities.

**Equity Units** and **Alumni and Advancement Offices** can play a role in securing this funding, while university admissions, scholarship offices and student wellbeing offices may be involved in administering funds. Funds may be delivered as a bursary, scholarship, or annual payment for care leavers to promote equitable participation in higher education. Direct payments are typically preferred by care leavers over vouchers as they offer more flexibility and avoid any potential perceptions of stigma.

#### Counselling

Many care leavers have experienced some form of counselling throughout their life. Our research showed that counselling was considered one of the best ways for care leavers to get through times of anxiety and stress.

#### **Careers advisers**

Careers advisers can be particularly important as care leavers often receive only limited careers advice and information prior to university. Careers advisers can also help students to plan an approach to balance their work and studies which takes financial and accommodation concerns into consideration. Advisers may assist with questions such as:

How can a student's income be supplemented while they are studying if they have rental payments and cost of living expenses?

What is a viable study plan that will develop the student's skills but will not overload the student?

#### **Culturally sensitive support**

Many care leavers in Australia are Aboriginal and/or Torres Strait Islanders. In addition to the support described in the sections above, most universities have Indigenous centres and programs which provide culturally sensitive information and support for Aboriginal and Torres Strait Islander people.

Staff within these units should be trained and regularly updated about the experiences and support available to care leavers, particularly given the over-representation of Indigenous children in out-of-home care.

Culturally sensitive support services can also be developed in consultation with agencies and organisations who operate in this space, for example the Aboriginal Child, Family and Community Care State Secretariat (AbSec) and KARI Aboriginal Resources Incorporated.

#### **Privacy and flexibility**

Special consideration forms, and other forms deigned to enable compensation for educational disadvantage during exams and assignments, often require personal information and paperwork. Care leavers may not be comfortable disclosing and documenting personal information within the higher education system, and may not have access to supporting documentation. Many care leavers experience severe to moderate anxiety which is tied to past experiences.

Our research shows that the flexibility of academic staff regarding assessment may provide support for, or bypass the additional workload of, applying for an extension. Students reported that being given flexibility provided a sense of understanding and autonomy, and reduced anxiety associated with academic workload.

#### Practice points

- Some care leavers described exhaustion from filling out paperwork throughout their lives. Where possible, it is best to reduce or avoid the burden of administrative tasks faced by care leavers.
- Some care leavers may be eligible for special consideration and/or disability support. At many universities, special consideration 'is intended for serious illness/injury/condition or extraordinary circumstances that has a *short term negative impact* on student's ability to sit/ attend/ participate/ complete the assessment task/s' (Medical Impact Statement, La Trobe University). Disability support may be provided for long-term illness, whereby a reasonable adjustment may be sought. Some students may be able to access disability support or an equivalent rather than lodging a special consideration application for multiple assessment tasks.

#### Anxiety and wellbeing

Care leavers may experience anxiety as a result of, for instance: insecure and unsafe living conditions; multiple placements with different families/or in shared accommodation; and experiences of abuse, neglect and violence. Anxiety may or may not be clinically diagnosed and documented. Anxiety and depression often manifest in unpredictable and unforeseeable ways, preventing students from attending class and completing set tasks. Drawing on our research, the following table shows how anxiety can affect the student experience and which supportive practices can help.

Special consideration and disability services 'So it's knowing my triggers and knowing that I might have a really bad week when an assignment's due, so making sure that my lecturers are aware of what's going on' ('Grace').	Is special consideration appropriate for the needs of care leavers? Disability services is another alternative for foreseeable illnesses that may have a long-term impact and require multiple extensions to assessment deadlines. Care leavers may not consider themselves entitled to disability services and this option should be pointed out to them.
Course content and triggers	Particular course content may trigger an unanticipated response for care experienced students. Creating awareness around potentially emotive or difficult course material can help students to prepare for their lessons ahead of time and avoid negative experiences. Social work, psychology and law courses could flag sensitive material at the outset of the course. This has the dual effect of raising awareness about the course material and the different backgrounds of students.
Temporarily unable to attend meetings or class in person 'I just need an extra 48 hours to get it done because you know I've had 48 hours where for no conceivable reason I have not been able to get out of bed' ('Grace').	<ul> <li>Many classes are recorded for students to listen to online. Students not wishing to attend meetings face-to-face could be offered alternative arrangements such as a phone call or Skype meeting.</li> <li>Counselling and stress management techniques (e.g. mindfulness meditation) and study management tools may be relevant to students.</li> <li>Students may be put in direct contact with someone from Counselling and Support Services, or the Student Development Advisers office.</li> <li>Assure students that they are entitled to use counselling services and may find them beneficial. Students may not be able to relate to particular counsellors and support workers and should be encouraged to meet with a variety of counsellors if necessary to find a good match.</li> <li>Peer groups and on-campus groups can provide support and foster relationships. These are often accessible through the student union.</li> </ul>

#### **Raising awareness and class discussions**

Care leavers may choose to share their care experiences in class in relation to course content. Our research shows that many care leavers who share their care experiences in group conversations in class, with peers, and with administrative staff, felt that they were not really understood or that they were pitied. Such experiences of pity may suggest a deficit approach of understanding care leavers. The idea that care leavers lack social and cultural capital, and are more 'needy' than other students, is contradicted by the resilience many care leavers demonstrate in overcoming significant barriers to participate in higher education.

#### Practice points

- It is beneficial for class conversations about care leavers to adopt a strengths-based approach. Many care leavers have life experience and knowledge which can contribute substantially to classroom discussions if a safe and understanding discussion environment is created.
- Misconceptions or discriminatory views about care leavers shared in class discussions should be engaged with critically.

#### Student voice:

'I think it's more life experience despite being the same age like you've done more, the rougher your life has been you get a little bit more mature and you get more understanding of how the world works' ('Dale').

## **Guideline references**

Australian Institute of Family Studies (AIFS). (2015). 'Children in care', CFCA Resource Sheet— June 2015, Australian Institute of Family Studies. URL: https://aifs.gov.au/cfca/publications/children-care

Australian Institute of Health and Welfare (AIHW). (2016). Child protection Australia 2014–15. Child welfare series no. 63. Cat. no. CWS 57. Canberra: AIHW.

Balvin Scholars. (2017). 'Balvin Scholars Program Philosophy', Available from: https://blavinscholars.umich.edu/article/philosophy

Berkeley Hope Scholars Program. (2017). Berkeley Hope Scholars Program. Available from: http://www.hope.berkeley.edu/

Lomax-Smith, J., Watson, L., & Webster, B. (2011) Higher Education Base Funding Review. Canberra: Department of Education, Employment and Workplace Relations.

Looked After Children & Care Leavers Project (2017). Available from: https://www.sheffield.ac.uk/schools/outreach-programmes/care-leavers-project

Harvey, A., Andrewartha, L. & McNamara, P. (2015a) A forgotten cohort? Including people from outof-home care in Australian higher education policy. Australian Journal of Education, 59(2), 182-195. doi: 10.1177/0004944115587529

Harvey, A., McNamara, P., Andrewartha, L., Luckman, M. (2015b) 'Out of care, into university: Raising higher education access and achievement of care leavers', Final Report, March 2015, Access & Achievement Research Unit, La Trobe University.

Peel, N. and Beckley, A. (2015) Children in Out of Home Care: Should we give them a Fair Go?. *Children Australia*, 40, pp 372-378 doi:10.1017/cha.2015.37

Phillips, L. A., Powers, L. E., Geenen, S., Schmidt, J., Winges-Yanez, N., McNeel, I.C., Merritt, L., Williamson, C., Turnter, S., Zweben, H. & Bodner, C. (2015) 'Better Futures: A validated model for increasing postsecondary preparation and participation of youth in foster care with mental health challenges', *Children and Youth Services Review*, 57: 50-59.

Sheffield University. (2017). 'Looked After Children and Care Leavers Project'. Available from: https://www.sheffield.ac.uk/schools/outreach-programmes/care-leavers-project

Who Cares? Trust. (2016). Propel. Available from: http://propel.org.uk/

# Section 5. Higher education guidelines for students who spent time in out-of-home care

## Thinking about uni?

#### If you're thinking about going to university in Australia then these guidelines are for you!

These guidelines are designed to show you *where* and *how* to start the application and enrolment process. We have focussed on services that people who have experienced out-of-home care are entitled to use to support their transition into higher education.

You might find these guidelines useful if you have ever been in foster care, kinship care, residential care, family group homes, home-based care, independent living, recently left care, transitioned to independence, or are a mature age student from a care background. Collectively, people from these groups are known as 'care leavers' within Australian higher education.

In this guide ...

In this guide you will find **short stories and quotes** from university students who are care leavers and **general information** (with links to webpages) about:

- What do I need to do first? The application and enrolment process
- Where will I stay? Year round accommodation
- How do people pay for uni? Financing your degree
- How will I pay my bills?
- What support does the student union offer?
- Shaping your future one step at a time
- What happens after my degree?
- University jargon
- Seven reasons to consider higher education.

We have provided links to the resources offered by different universities to show examples of the kinds of support available and where you can find these online. Along the way we have included **peer advice** from care experienced students who may have been through a similar situation to you!

#### STUDYING AT UNIVERSITY

'Where you have been does not define who you are, there is no you, you are who you are in the moment and you can choose who you are in the moment. Yeah you might be a bit behind from other families, but everybody's got their struggles and you are no less important and worthwhile than any other person, and if you want to do something just do it' ('Casey', aged 27).

There is also a brief **glossary** at the end of these guidelines which explains common higher education terms and jargon.

Disclaimer

The information contained in these guidelines is of a generic nature only. You must not rely on the information in these guidelines as an alternative to seeking advice from an appropriately qualified professional and/or making your own enquiries to other higher education institutions to confirm the accuracy of the information provided. The information contained in this publication is current as at the time of the publication (March 2017). The University reserves the right to make changes without notice, at any time in its absolute discretion, including but not limited to varying these guidelines. *These guidelines must not be publised, quoted or disseminated to any other party without the University's prior written consent*. To the extent permitted by law, the University does not accept responsibility for any loss or damage occasioned by use of any of the information contained in this publication.

#### **IS UNIVERSITY FOR ME?**

University is a place to make new friends, develop skills and knowledge, plan for your future, and enhance your career opportunities upon graduation. The physical location of your university (buildings, lecturers and tutors) is referred to as the university **campus**. This is where you will attend class if you are an on-campus (face-to-face) student.

University is different from school in terms of study and assessment, hours you spend on campus (at the university), time spent in class, and relationships with teachers (or lecturers and tutors). For instance, you can choose which classes you attend to suit your own schedule. Universities encourage independent study, group study and peer learning and provide a number of support services to suit different needs.

There are many different pathways into university and different ways to apply depending on your stage in life. Some people complete bridging programs which enable them to apply for the course they are interested in, while some people apply to university after finishing high school.

A diverse range of courses are offered by Australian universities, including Law, Nursing, Education, Commerce, Science, Health Sciences, Engineering, Psychology, Sociology, Gender Studies, Medicine, Creative Arts, Architecture, Urban Planning, Accounting and Social Work. Undergraduate courses may take between 3 and 4 years to complete, during which time students may choose to work casually or part-time, seek government assistance and other forms of financial assistance.

#### What do I need to do first? The application and enrolment process

## Finding the right pathway into university for you

Applying for university can be tricky. As a first step many universities offer pre-application help and advice to future or prospective students through careers and employment contacts listed online. For instance, <u>Federation University of Australia</u> has a contact number for people thinking of applying. The Queensland University of Technology has an <u>introductory page</u> for people considering study, as does <u>Western Sydney University</u>. La Trobe University provides prospective or <u>'Future Students'</u>, including <u>Indigenous Students</u>, with contact names and a phone number for specific questions.

#### Australian Tertiary Admission Rank (ATAR) / Overall Position

The ATAR, or, in Queensland, the 'Overall Position', is the primary criterion for entry into most undergraduate university degrees in Australia. The ATAR and Overall Position are measures of secondary school academic achievement that help institutions to rank applicants for selection to tertiary education courses.

#### 'Alternative pathways'

There are many different or 'alternative' pathways into higher education, which include: bridging and enabling programs, vocational education and training studies, school/principal recommendation schemes, and early starter programs.

#### Bridging Programs provide one pathway to university

'I lived in a number of different foster homes. I did really, really well at school but ... I didn't finish my VCE. I worked for a number of years, then got the urge to go to university. A friend told me about a Bridging Program and said it would be good to give it a go to test myself out. I enrolled, and found it straightforward and easy to get through.

I guess having had my last experience of education so long ago and I was so disengaged from my education when I was there it was a little bit nerve wracking but there weren't any barriers or difficulties.

There was no automatic guarantee in to uni but fortunately I was offered a place ... so that was my pathway into uni, it was the Bridging Program. I'd recommend a Bridging Program to other care leavers who have been out of school for a while' ('Samantha').

'Kurt' explains that as a mature age student he applied to university through an 'alternative' entry scheme:

#### Alternative entry into university

'I essentially enrolled in night school to try and improve my grades ... during that time I heard about industrial design, and I thought yeah that's the ticket for me. So I did chemistry and English, I was about to do Physics, but I heard that it wasn't necessary to do industrial design. Then I applied to do industrial design, my grades weren't terribly good, but because I was older than 25 I could get in through alternate entry, and I did. I drew up a big portfolio of ideas and I got in through that. I did the undergrad, thought yeah this is awesome I'm going to do the grad dip' ('Kurt', aged 49).

Here is some more information about alternative pathways into university:

- School/principal recommendation schemes provide students in school with early offers to attend university. Offers are made to students before final exam results. These schemes are available to students from selected partner schools.
- Special entry and educational access schemes help universities to take personal circumstances into consideration when considering university applications. If your education was affected as a result of disadvantage you may be eligible to apply. These schemes have different names in different states. For instance, in Victoria it is called the <u>Special Entry</u> <u>Access Scheme</u> (SEAS), and in New South Wales, Queensland and the Australian Capital Territory it is called the <u>Educational Access Scheme</u> (EAS). You will also need to apply for your preferred university course. These application processes can be found online through the Universities Admissions Centre which can be accessed online at: www.uac.edu.au

Special entry and educational access scheme applicants can take the <u>Special Tertiary</u> <u>Admissions Test</u> (STAT). It is a single application that is sent to all the courses you've applied to. You can choose a category through which to apply. They provide four categories, including Difficult Circumstances, and Disability or Medical Condition. For example, Category 1: <u>Personal Information and Location</u>, includes 'Recognition as an Indigenous Australian', which applies to applicants of Australian Aboriginal and/or Torres Strait Islander descent OR identify as a person of Australian Aboriginal and/or Torres Strait Islander heritage.

- Enabling programs are pre-bachelor degrees which prepare you for a university course. Enabling programs are also referred to as <u>Bridging</u> Courses and <u>Tertiary Enabling programs</u> depending on the education provider. Preparation or bridging programs give you a taste of university and prepare you with the skills necessary to do well in your university degree. Many programs are provided at no cost and vary in duration, from several weeks to a year or more. Bachelor degrees are undergraduate university courses, for example, a Bachelor of Science (B.Sc.) or Bachelor of Arts (B.A.).
- You can also apply to university on the basis of partial or completed <u>Vocational Education and</u> <u>Training (VET)</u> studies. **VET** studies are skills-based and can often be used as a stepping stone to higher education. VET courses are offered through Registered Training Organisations (e.g. TAFE institutes and private providers). They are also offered through some secondary schools, industry bodies and adult and community education centres.

#### Look out for:

- Many universities have **university handbooks** with essential information about courses on offer.
- Student webpages and Frequently Asked Questions (FAQ) webpages may have helpful hints.
- Some universities offer different options for entry without an ATAR. For instance, Federation College's ASPIRE pathway program integrates <u>Victorian Certificate of Applied Learning</u> (VCAL), Foundation Access Studies (FAST) and <u>Embedded Academic Transition (EAT)</u> units.

# Getting from high school to university

#### Support from school careers advisors

'I always wanted to go to university. School gave me lots of encouragement especially the careers advisor. She showed me the different options for applying to university. Saved me a lot of time and effort. A lot of it I did do on my own, but the careers advisor let me know you can apply for this bonus point scheme, for scholarships and all that kind of stuff ... Out of school there was a lot going on as well, so I did see her once a week or once a fortnight ... she helped me a lot' ('Brooke').

#### Leaving care and going to university

'I applied to go straight into university after high school. There's so much that you're thinking about, well that I was thinking about when I was leaving care you know because my birthday's in the middle of the year as well and DHS were like your payment gets cut off in May and you've still got the rest of the year to go. So I wasn't thinking "yay I'm going to go to university next year that's going to be great fun", you know it was kind of like it was one of the last things that I was thinking about, and I was only thinking about it because I had the foster parents who were encouraging that' ('Kelly').

Application processes for higher education enrolment with an Australian Tertiary Admission Rank (ATAR) vary depending upon what state you live in and what higher education institution you are aiming for. For instance, if you are in NSW or the ACT you can apply through the <u>Universities</u> <u>Admissions Centre (UAC)</u>. The application process online will walk you through each of the required steps. Ways of applying in different states are as follows:

- New South Wales and ACT (University Admissions Centre (UAC)
- Northern Territory and South Australia: South Australian Tertiary Admissions Centre (SATAC) (<u>www.satac.edu.au</u>)
- Queensland: Queensland Tertiary Admissions Centre (QTAC) (www.qtac.edu.au)
- Tasmania: University of Tasmania (<u>www.utas.edu.au</u>)
- Victoria: Victorian Tertiary Admissions Centre (VTAC) (<u>www.vtac.edu.au</u>)
- Western Australia: Tertiary Institutions Service Centre (TISC) (<u>www.tisc.edu.au</u>)

**Prerequisites** for higher education vary depending on the course and university you choose, and previous studies will be taken into consideration. Some courses will allow you to submit an application **directly** (through 'direct admission') to the higher education provider, while others require applications through your relevant **Tertiary Admissions Centre** (as listed above). To find options for undergraduate domestic students and international students visit your *university admissions webpage*.

If you did receive an ATAR score but are worried it will not be high enough to get you into your preferred course at university, you may wish to apply for an educational access scheme, or sit the <u>uniTEST</u>. Be sure to check that the university you wish to apply to will accept uniTEST scores. For instance, Flinders University and the University of Canberra will consider uniTEST scores.

# Aboriginal and/or Torres Strait Islander Students

#### Support from Indigenous centres

'I am 18 years old and in my first year at university studying a Bachelor of Exercise Science and Behavioural Science. I get a lot of support from the Indigenous unit on campus. As an Aboriginal student I can access free tutors, organise keystones to success and if I want to go in and I don't understand something I can have someone to talk to and we sit there for an hour and work that out. I'm working towards becoming a sport psychologist or exercise scientist, because I'd like to help people improve their skills in the sporting world' ('Dale').

In addition to the pathways we have described in the sections above, many universities have Indigenous student support units and programs which provide culturally sensitive information and support for Aboriginal and Torres Strait Islander people thinking of applying to university. For example:

- The Oodgeroo Unit at Queensland University of Technology provides Aboriginal and Torres Strait Islanders with academic, personal and cultural support, and is also a research unit. The Oodgeroo webpage contains information for Aboriginal and Torres Strait Islander people considering higher education, including through the <u>Centralised Assessment Selection</u> <u>Program</u> (CASP). CASP is an alternate entry pathway to university study which considers prior learning and other personal achievements, and may provide students with a bursary for study or living costs. CASP also offers units of study in: Indigenous history, culture and current issues.
- Through their **Aboriginal and Torres Strait Islander (ATSI)** webpage, Western Sydney University offers the <u>School of Medicine Indigenous Program</u>. They also offer Aboriginal and Torres Strait Islander students the option of applying through their Alternative Entry program. The application process involves submitting your application online and then participating in a two day program.
- Deakin University's **Institute of Koorie Education (IKE)** offers Community-Based Education programs to all Indigenous Australians across the nation. Courses are available in faculty undergraduate and postgraduate studies across the arts, education, health, business and law.

Indigenous units may also offer cultural programs and events. For instance, Indigenous student service units at La Trobe University offer the <u>Sport Occupations and Aspiration Raising</u> camp (SOAR), which is a free four-day camp that introduces Year 9 and 10 Indigenous students to university and sports occupations. <u>Information evenings</u> for prospective Aboriginal and Torres Strait Islander Students are a good way to find out more about university, how to apply and what course you are interested in. Support is also offered to culturally and linguistically diverse (CALD) students at most universities. <u>CALD</u> support workers can assist with information about courses and pathways into employment.

#### Where will I stay? Year round accommodation

Some universities offer residential ('rez') services to their students which includes on-campus accommodation. Many universities have an accommodation website containing information about how many days per year accommodation is available (e.g. 365 days per year) and how you can apply. Applications received before a certain date may be given special consideration and many universities **recommend applying early** to secure a place. In Victoria, VTAC provides a <u>list of universities</u> offering accommodation.

The University of Newcastle, for example, has a subsided accommodation program designed for care leavers called <u>Live, Learn, Grow</u>.

Accommodation may be subsidised, or <u>Accommodation Services Access Scholarships</u> may be available.

#### LEARNING LIFE SKILLS

'Start look at getting skills in all factors of life, so if you are living out of home in accommodation share housing or in a place by yourself, try cooking classes. You have to learn how to cook, I didn't know how to cook! Also, I feel exercise and organisational skills are quite an important thing' ('Chris', age 23).

## How do people pay for uni? Financing your degree

## HECS-HELP (previously known as HECS)

After you have applied for your course you can organise finance for your degree. The costs of higher education are substantial. However, it is important to remember that *no* tuition money is required up front.

- If you are an Australian citizen the **Commonwealth Government** pays for part of your course (through what is called a 'Commonwealth Supported Place', CSP), with your tuition fees being initially covered by an income-contingent loans scheme, called **HECS-HELP** (Higher Education Contribution Scheme-Higher Education Loan Programme). You can get a HECS-HELP application form from your university. You only have to pay the HECS-HELP loan back once you start earning a certain income, and the debt is drawn from your salary when you begin work and earn over a certain amount. More information about applying for HECS-HELP is available from the Australian Government <u>Study Assist website</u>.
- Before gaining Commonwealth assistance you must be enrolled with an <u>eligible</u> <u>Australian education provider</u>.

# Scholarships

#### Applying for scholarships

Towards the end of year 12 'Olivia' (age 21) applied for university using the Victorian Special Entry Access Scheme and was accepted into a Bachelor of Sport Science. She also applied for an equity scholarship available through the university. She says:

'I enquired on their page, I just said "Hey what scholarships do you offer?" and they sent me an email saying that I could be eligible'.

There are different types of funding to support higher education participation. As you scroll through the online Tertiary Admissions Centre relevant to your location you can find different scholarship and bursary opportunities. Scholarships provide financial support to students while they are studying. There are many different types of scholarships, which can be based upon work experience, academic achievement, or life experience. These may include: Institutional Access and Equity Scholarships and Commonwealth Scholarships.

Some universities offer more than *80 different kinds* of scholarships, including: <u>work scholarships</u>, <u>accommodation scholarships</u>, <u>equity scholarships</u> and scholarships for <u>Indigenous Australians</u>. Indigenous Australian are also offered supported through the <u>Aspiration Initiative</u> which offers scholarships for undergraduate study at Australian universities as well as postgraduate scholarships for study in Australia and overseas. Many universities offer their own scholarships for Indigenous Australian. For instance, Western Sydney University offers the <u>Pathways To Dreaming Indigenous</u> <u>Achievement Scholarship</u>.

- Some scholarships are one-off payments but others can be for the duration of the degree.
- Particular paperwork may be required during the application process, but if you do not have the paperwork you may not be excluded. Make sure you get in contact with someone at the university to see what options they can provide for you.
- Many of these scholarships prioritise students most in need (i.e. ranked criteria is based on financial disadvantage criteria), which means that they are designed for students who have faced hardships or will need financial support.

#### Scholarship funds do not have to be repaid.

#### Targeted funding for care leavers

Applying for targeted funding (funding *only* available to care leavers) is a great way to ensure you are receiving all the support to which you are entitled. There are some <u>bursaries</u> and <u>scholarships</u> specifically for people who have been in out-of-home care and these are often listed under the 'scholarships' page on your higher education provider webpage. For instance, Western Sydney University offers the NSW Department of Family and Community Services Out-of-Home Care Pathways <u>Scholarship</u>.

Does the higher education institution you are considering have a website specifically for people who have experienced out-of-home care? <u>Swinburne University</u>, <u>Federation University</u>, and <u>La Trobe</u> <u>University</u> all offer access scholarships for care leavers. Some universities have gathered together information online regarding the resources available to assist <u>care leavers</u>. A search of the term 'care leaver' or 'out-of-home care' on your university webpage should direct you to the relevant page.

## How will I pay my bills?

#### **ORGANISING FINANCES**

'Apply for all financial aid that you can including Centrelink, equity scholarships, also try and get a job. Apply for as many jobs as you can so you get a part time job, and then yeah kind of get money, because that's the one thing that's really going to mess you up if you don't have it, especially the stress of bills and accommodation etc' ('Chris', age 23).

#### Government Assistance: Youth Allowance and Student Start-up Funding

Many students qualify for the Youth Allowance which is a means-tested payment for full-time students and Australian Apprentices generally aged 16 to 24 years old. Youth Allowance may be paid for the duration of your degree on a fortnightly basis. You can apply by going to this webpage:

#### http://studyassist.gov.au/sites/studyassist/studentincomesupport

If you are eligible to receive student payments you may also be eligible to receive the **Student Start-up Scholarship** or **Student Start-up Loan** and the **Relocation Scholarship**.

#### Youth Allowance and Start-up Scholarships do not have to be repaid.

#### Loans

Interest free **student loans** often ranging between \$2,000 and \$5,000 may be available through the university or student union to support essential living expenses like the cost of computers, textbooks, practicum costs, rent or bond, utilities, and food. Student Loans may also be referred to as 'Financial Schemes', 'Financial Assistance', 'Bursaries', or 'Student Grants'.

#### **Pathway Plan**

A **pathway plan** or **leaving care plan** sets out your aims and goals for the next phase of your life after you turn 18. The plan might be written with a case worker or other members of your support network and may contain important information regarding living arrangements, employment and income, further education and, if relevant, cultural support. Plans may include financial support up to the age of 21 in Victoria, and 25 in other states, with current variations in Queensland and the ACT. This plan may include the <u>Transition to Independent Living Allowance</u> (TILA) which is a one-off payment of up to \$1,500 for young people aged 15–25 who are moving from care and who qualify for independent status under Centrelink guidelines.

A solid leaving care plan can help facilitate your educational goals. If you do not yet have a plan, why not reach out to your support network and start creating one?

# To find out more about the funding that is available to you, contact your preferred higher education institution.

Students can also contact <u>Equity and Diversity</u>, and Widening Participation offices to see if out-ofhome care projects/supports specific to students from out-of-home-care are available.

#### What support does the student union offer?

Many universities have a student union which offers a range of services that students are entitled to use, including <u>text book vouchers</u>, <u>peer support and study skills programs</u>, free <u>legal support</u>, <u>Clubs</u> and <u>Societies</u> and bookstores. *Unions usually operate independently of the University to privately and confidentially support your needs*. Their services may include the following:

*Financial Planning and Aid* - This service may also be called 'Financial Counselling' and can help you with budgeting, managing different sources of income, and accessing funding you are entitled to. Good financial planning is important!

**Student groups and societies -** Making connections and building networks is streamlined by the student groups and societies on campus. Simply sign up to the group that you are interested in. This can be a great way to make lasting friendships.

**Advocacy** - This is the process through which students are supported by staff working with the student union to address an academic, administrative or welfare related issues (e.g. special consideration, assessment issues).

There are also groups outside the university sector that could provide support or advocacy e.g. CREATE for under 25s and CLAN for older Care Leavers.

*Emergency accommodation -* This service is designed for emergencies like not having a place to stay. Clothing and food may also be provided.

Travel and Cost of Living support - Including subsidies for travel to and from university.

#### Shaping your future one step at a time

#### Deferring and changing courses

'I applied to go straight into university after high school. I got into university to do a Bachelor of Nursing and I relocated there for a month. I wasn't really coping that well with it so I deferred and worked for 2 years and then tried a diploma of nursing and figured it wasn't for me'.

Kelly decided to do a different degree to the one she was enrolled in, which was a tough decision to make. However, she is now happily enrolled in a Bachelor of Arts and receives a lot of support from her housemates (*'Kelly'*).

Studying can sometimes be stressful and sometimes you just need to talk to someone, find out what your options are, or be pointed in the right direction! Most university support services or welfare services are free and intended for you to use if and when you need them. Support services include mentoring and counselling.

Mentoring programs are offered by many universities and are a great way to settle into university life. Western Sydney University offers a <u>MATES</u> program for new students. MATES stands for Mentoring and Transition Equals Success. La Trobe University offers a <u>Connect Mentor Program</u>.

Counselling is a confidential and private service for all students and focuses on social and emotional wellbeing. Individual counselling may be offered in person, <u>online</u>, and via telephone. Self-help resources are also available for dealing with stress and anxiety, relationship concerns and study habits.

Centre for Higher Education Equity and Diversity Research latrobe.edu.au/cheedr

#### Finding a counselling service

'If I had to pick one thing that I would say to a student who was say 23 and had lived in care and was now going to go to uni, I'd say find a counselling service and be prepared that the emotions related to that experience which may not be apparent in the everyday here and now might come up, and you will then have help dealing with them' ('Lauren', age 44).

Culturally sensitive support services are offered by most universities in Australia. Services range from counselling to academic support. Some universities have an Indigenous counsellor for students to access, while other universities offer support through specific Aboriginal and Torres Strait Islander schools, <u>units</u> and <u>centres</u>.

Most universities offer careers service and support. For example Federation University offer <u>Student</u> <u>Careers and Employment Services</u>, while the University of New South Wales have a <u>Careers and</u> <u>Employment</u> webpage which provides information about career planning and professional development advice. Careers advice and course advice is offered to all students to help you find the right course for you.

Mature Age students (over 21 years of age) are entitled to access university support services, and are also supported by the university through special entry access schemes and bridging programs that recognise prior work experience as well as academic qualifications.

*Study Groups (also called Peer Assisted Study)* can be a great way to learn course material and make new friends. Many universities have study groups already set up for students to attend. It often pays to look for these groups in the 'Clubs and Societies' webpage of your university.

#### Putting time into study

'The amount of time put in equals marks ... When I started going to uni the year I passed was when I studied heaps. One of my big things was just to go to everything ... literally going to every single lecture, that is really, really important and I think a lot of students really don't understand that' ('Chris', age 23).

#### Look out for:

• <u>Student advisors</u>, <u>learning advisors</u> and <u>mentors</u> are there to provide information and direction.

#### What happens after my degree?

#### Considering postgraduate study

'I took some time off between school and university to figure out exactly what I wanted to do. So I started university as a mature age student. I wasn't sure I would be able to do it but I surprised myself and went on to complete a Masters degree with the support of my supervisor, after that I was then able to enrol in a PhD. I thought I'd like to get back into research, that's when I approached my supervisor, and I got in ... And yeah I'm loving it' ('Kurt', age 49).

Once you have settled into your higher education course it will be time to think about the next stage of your life. Like 'Kurt' you may want to undertake more study, especially if you have high grades! After a Bachelor degree, you can often apply to study Honours or a new postgraduate coursework or research degree, such as a Masters degree or a graduate diploma.

You may have met with a career planner when you first enrolled to find out which subjects suit your interests and career goals. The careers office is a good place to seek advice and start your planning for life after your studies. Many education providers have a careers counsellor, a careers and employment team, or a careers mentor scheme available. For instance, the <u>careers team</u> at La Trobe University offer advice about <u>career options</u>, provide interview practice and help students create a career pathway (see also careers advice and course advice in Section 5).

## Aiming high

'I think aiming high is really important despite the circumstances, you might get knocked down a lot if you do have a very demotivating carer or a negative carer but I think striving for your personal best is the most important thing' ('Brooke', age 18).

# University jargon

Year 11 & 12 Certificates

Australian Tertiary Admission Rank (ATAR)	ATAR is a measure of secondary school academic achievement that helps institutions to rank applicants for selection to tertiary education courses.
Higher School Certificate (HSC)	Secondary school students who successfully complete senior high school level studies (Years 11 and 12 or equivalent) in New South Wales are awarded a Higher School Certificate (HSC).
Northern Territory Certificate of Education (NTCE)	Secondary school students who successfully complete senior high school level studies (Years 11 and 12 or equivalent) in the Northern Territory are awarded a Northern Territory Certificate of Education (NTCE).
Queensland Certificate of Education (QCE)	Secondary school students who successfully complete senior high school level studies (Years 11 and 12 or equivalent) in Queensland are awarded a Queensland Certificate of Education (QCE).
South Australian Certificate of Education (SACE)	Secondary school students who successfully complete senior high school level studies (Years 11 and 12 or equivalent) in South Australia are awarded a South Australian Certificate of Education (SACE).
Tasmanian Certificate of Education (TCE)	Secondary school students who successfully complete senior high school level studies (Years 11 and 12 or equivalent) in Tasmania are awarded a Tasmanian Certificate of Education (TCE).
Tertiary Admissions Centres (TAC)	Tertiary Admissions Centres play a major role in university admissions processes. They receive applications and forward them to the universities. Universities then decide who they will offer positions to.
Universities Admissions Centre (UAC)	The Universities Admissions Centre (UAC) processes applications for admission to most undergraduate courses at participating institutions (in NSW and the ACT).
Victorian Certificate of Education (VCE)	Students who successfully complete high school level studies (year 11 and 12 or equivalent) in Victoria are awarded a Victorian Certificate of Education or VCE.
Western Australian Certificate of Education (WACE)	Secondary school students who successfully complete senior high school level studies (Years 11 and 12 or equivalent) in Western Australia are awarded a Western Australian Certificate of Education (WACE).

Other terms you might come across

Admissions	The Admissions office will assist you with your enrolment and application process, they may also be called the 'Enrolment Office'.
Commonwealth Supported Place (CSP)	A Commonwealth supported place (CSP) is a <i>government subsidised</i> <i>higher education</i> enrolment. CSPs are available to domestic students and most undergraduate students studying at university are enrolled in a CSP. This simply means the Commonwealth Government pays for part of your course, which is the case for most domestic students in Australia.
Higher Education Contribution Scheme- Higher Education Loan Programme (HECS- HELP)	Higher Education Contribution Scheme-Higher Education Loan Programme (previously known as HECS) is a loan scheme which means that no tuition money is required up front. The loan can be paid back over time, after you begin work and start to earn a salary over a certain amount.
Mentoring	Mentoring is when you receive advice from a person with knowledge and/or experience in a particular area. This may be a formal or informal arrangement.
Means Tested	Some forms of financial assistance may be 'means tested' which means that your family income will be taken into consideration. However, many people leaving care become independent upon turning 18, as a result of this independence you will be entitled to many forms of means tested financial support.
Open days	This is a specific day when potential and new students are invited on campus to explore the university. There are often information booths set up so that you can ask questions and get brochures.
Outreach events	Outreach events often involve different activities and presentations and are designed to give people extra information about university. They can be held on campus, at school or at community centres.
Pre-enrolment	Pre-enrolment refers to the period in which you are thinking about university and how you will apply.
Student Finance	Student finance refers to financial support for students and the different kinds of funding available to students.
Student Advisors	A Student Advisor is a person who provides advice and direction. Students Advisors can be your first point of contact at the university.
Support Services /Student Welfare	Support services are often offered through the student union and may include counselling and financial advice through programs and one on one meeting sessions with staff.

#### Seven reasons to consider higher education

- 1. Having greater access to a wide variety of jobs after graduation, and having the capacity to earn more money
- 2. Living independently and being in control of your future
- 3. Gaining confidence through your experience and achievements
- 4. Finding clubs and societies to suit your personal and political interests
- 5. Learning and pursing your own interests in a welcoming and supportive space
- 6. Having the chance to make lifelong friends
- 7. Choosing the town or city where you want to live.

# References

- All-Party Parliamentary Group for Looked After Children and Care Leavers. (2012). *Education matters in care: a report by the independent cross-party inquiry into the educational attainment of looked after children in England* (London: Who Cares Trust).
- Australian Bureau of Statistics. (2013). Young adults: then and now. Retrieved 12 June 2014, from http://www.abs.gov.au/AUSSTATS/abs@.nsf/Lookup/4102.0Main+Features40April+2013#liv ingar
- Australian Council of Social Services. (2014). Poverty in Australia Report 2014. Strawberry Hills, NSW: Author.
- Australian Institute of Family Studies. (2017). Beyond 19: the longitudinal study of leaving care. Retrieved 10 February 2017 from https://aifs.gov.au/projects/beyond-18-longitudinal-studyleaving-care
- Australian Institute of Health and Welfare (AIHW). (2007). *Educational outcomes of children on guardianship or custody orders: a pilot study,* Child Welfare Series no. 42. Cat. no. CWS 30 (Canberra: Australian Government).
- Australian Institute of Health and Welfare (AIHW). (2011). Educational outcomes of children on guardianship or custody orders: a pilot study, Stage 2, Child Welfare Series no. 49. Cat. no. CWS 37 (Canberra: Australian Government).
- Australian Institute of Health and Welfare (AIHW). (2016). Child Protection Australia, 2014-2015. Child Welfare Series 63. Canberra: AIHW.
- Australian Institute of Family Studies (AIFS). (2015). 'Children in care', CFCA Resource Sheet— June 2015, Australian Institute of Family Studies. Retrieved 23 January 2017 from: <u>https://aifs.gov.au/cfca/publications/children-care</u>
- Beauchamp, T. (2014). A strong future for young people leaving out-of-home care: UnitingCare
   Children Young People and Families position paper on leaving care and aftercare. New South
   Wales: UnitingCare Children, Young People and Families.
- Blavin Scholars. (2016). Blavin Scholars Program. Retrived from http://blavinscholars.org/blavinscholars-program
- Cameron, C., Connelly, G., & Jackson, S. (2015). *Educating children and young people in care: Learning placements and caring schools*. London: Jessica Kingsley Publishers.
- Cashmore, J., Paxman, M., & Townsend, M. (2007). 'The educational outcomes of young people 4–5 years after leaving care: an Australian perspective', *Adoption & Fostering*, 31(1), 50-61, doi: 10.1177/030857590703100109
- Courtney, M., &Dworsky, A. (2005). *Midwest evaluation of the adult functioning of former foster youth: outcomes at age 19* (Chicago: Chapin Hall Center for Children at the University of Chicago).

- Courtney, M., Dworsky, A., Brown, A., Cary, C., Love, K., & Vorhies, V. (2011). *Midwest evaluation of* <u>the adult functioning of former foster youth: outcomes at ages 26 (Chicago: Chapin Hall at</u> the University of Chicago).
- Courtney, M., Dworsky, A., Lee, J., & Raap, M. (2010). Midwest evaluation of the adult functioning of former foster youth: outcomes at ages 23 and 24. Chicago: Chapin Hall at the University of Chicago.
- CREATE Foundation. (2006). Report Card on Education 2006 (Sydney: CREATE Foundation).
- Creed, P., Tilbury, C., Buys, N., & Crawford, M. (2011). 'The career aspirations and action behaviours of Australian adolescents in out-of-home-care', *Children and Youth Services Review, 33*, 1720–1729, doi: 10.1016/j.childyouth.2011.04.033
- Deloitte Access Economics. (2016). Raising our children: guiding young Victorians in care into adulthood. Report commissed by Anglicare Victoria. Retrieved 3 February 2017 from http://thehomestretch.org.au/site/wp-content/uploads/2016/08/Raising-Our-Children\_Guiding-Young-Victorians-in-Care-into-Adulthood.pdf
- Department for Business Innovation and Skills. (2014). National strategy for access and student success in higher education. (London: Department for Business Innovation and Skills).
- Department for Education. (2014). *Children looked after by local authorities in England: Guide to the SSDA903 collection 1 April 2014 to 31 March 2015*.
- Department of Education Employment and Training (DEET). (1990). *A Fair Chance for All: national and institutional planning for equity in higher education* (Canberra: Australian Government Publishing Service).
- Department of Social Services. (2014). Transitioning to independence from out of home care: Discussion paper. Retrieved 11 June 2014, from http://www.dss.gov.au/ourresponsibilities/families-and-children/publications-articles/transitioning-to-independencefrom-out-of-home-care-discussion-paper?HTML
- Fernandez, E. (2008). 'Unravelling emotional, behavioural and educational outcomes in a longitudinal study of children in foster-care', *British Journal of Social Work*, 38, 1283-1301, doi: 10.1093/bjsw/bcm028.
- Frederick, J., & Goddard, C. (2010). 'School was just a nightmare: childhood abuse and neglect and school experiences', *Child and Family Social Work, 15*, 22-30, doi: 10.1111/j.1365-2206.2009.00634.x
- Gale, T., & Parker, S. (2013). Widening participation in Australian higher education. London: Higher Education Funding Council of England and the Office for Fair Access.
- Garbarino, J. (1977). The human ecology of child maltreatment: a conceptual model for research. *Journal of Marriage and Family, 39*(4), 721-735.
- Harvey, A., Andrewartha, L. & McNamara, P. (2015). A forgotten cohort? Including people from outof-home care in Australian higher education policy. *Australian Journal of Education, 59*(2), 182-195. doi: 10.1177/0004944115587529
- Harvey, A., Burnheim, C. & Brett, M. (eds). (2016). Student Equity in Australian Higher Education: Twenty-five Years of A Fair Chance for All, Singapore: Springer.

- Harvey, A., McNamara, P., Andrewartha, L., & Luckman, M. (2015). *Out of care, into university: raising higher education access and achievement of care leavers* (National Centre for Student Equity in Higher Education (NCSEHE), Curtin University). Retrieved 29 June 2015 from https://www.ncsehe.edu.au/research/research-reports/
- Home Stretch. (2016). The Home Stretch Campaign. Retrieved 24 January 2017 from http://thehomestretch.org.au/about/
- Jackson, S., Ajayi, S., and Quigley, M. (2005). *By Degrees: going to university from care,* (London: Institute of Education, University of London).
- Jackson, S., & Cameron, C. (2012). *Final report of the YiPPEE project. Young people from a public care background: pathways to further and higher education in five European countries* (London: Thomas Coram Research Unit, University of London).
- Jackson, S., & Cameron, C. (2014). *Improving access to further and higher education for young people in public care: european policy and practice* (London: Jessica Kingsley Publishers).
- Johnson, G., Natalier, K., Mendes, P., Liddiard, M., Thoresen, S., Hollows, A., & Bailey, N. (2010). *Pathways from out-of-home care.* (Melbourne: Australian Housing and Urban Research Unit).
- Jurczyszyn, R., & Tilbury, C. (2012). 'Higher and further education for care leavers: a road less travelled', *Developing Practice: The Child, Youth and Family Work Journal*, 33, 10-22.
- Legal Center for Foster Care and Education. (2008). *Foster care & education Q & A. Federal laws that increase educational opportunities for older youth in out-of-home care* (Chicago: American Bar Association and Casey Family Programs).
- McDowall, J. J. (2013). Experiencing out-of-home care in Australia: the views of children and young people. CREATE Report Card 2013. Sydney: CREATE Foundation.
- McFarlane, K. (2010) 'From care to custody: young women in out-of-home care in the criminal justice system', *Current Issues in Criminal Justice*, *22*(2), 345-353.
- McLachlan, R., Gilfillan, G., & Gordon, J. (2013). Deep and persistent disadvantage in Australia. Productivity Commission Staff Working Paper. Canberra, Australia: Productivity Commission.
- Mendes, P. (2014). Leaving care, or left alone? How not to fail young people transitioning from outof-home care. Retrieved 11 June 2014, from http://www.abc.net.au/religion/articles/2014/02/13/3944328.htm
- Mendes, P. (2009). 'Young people transitioning from state out-of-home care: Jumping hoops to access employment', Family Matters, 83: 32-38.
- Mendes, P., Michell, D., & Wilson, J. Z. (2014). 'Young people transitioning from out-of-home care and access to higher education: a critical review of the literature', *Children Australia*, 39(4), 243-252.
- Mendis, K., Gardner, F., & Lehmann, J. (2014). 'The education of children in out-of-home care', Australian Social Work, 68(4), 483-496, doi: 10.1080/0312407X.2014.963134
- Michell, D., Jackson, D., and Tonkin, C. (eds.). (2015). *Against the odds: care leavers at university* (Elizabeth, South Australia: People's Voice Publishing).

- National Network for the Education of Care Leavers (NNECL). (2013). 'Supporting Care Leavers in FE and HE: A guide for those taking on the role of Care Leaver Contact', National Network for the Education of Care Leavers, 1<sup>st</sup> National Edition. Retrieved 23 January from: http://www.nnecl.org/resources/nnecl-supporting-care-leavers-in
- Office for Fair Access. (2015). Topic briefing: Care leavers. Retrieved from
  <a href="https://www.offa.org.uk/universities-and-colleges/guidance-and-useful-information/topic-briefings/topic-briefing-care-leavers/">https://www.offa.org.uk/universities-and-colleges/guidance-and-useful-information/topic-briefings/topic-briefing-care-leavers/</a>
- Pecora, P., Kessler, R. C., Williams, J., O'Brien, K., Downs, A. C., English, D., & Holmes, K. (2005). *Improving family foster care: findings from the Northwest Foster Care Alumni Study* (Seattle: Casey Family Programs).
- Pecora, P., Williams, J., Kessler, R. J., Downs, A. C., O'Brien, K., Hiripi, E., & Morello, S. (2003). Assessing the effects of foster care: early results from the Casey National Alumni Study (Seattle: Casey Family Programs).
- Phillips, L. A., Powers, L. E., Geenen, S., Schmidt, J., Winges-Yanez, N., McNeel, I.C., Merritt, L., Williamson, C., Turnter, S., Zweben, H. & Bodner, C. (2015). Better Futures: A validated model for increasing postsecondary preparation and participation of youth in foster care with mental health challenges. Children and Youth Services Review, 57, 50-59.
- Queensland State Government (2015). Next Step Survey. Retrieved 20 February 2017, http://education.qld.gov.au/nextstep/nextstep.html
- Queensland Tertiary Admission Centre (QTAC). (2017). Educational Access Schemes, Retireved 7 February 2017 from, http://www.qtac.edu.au/applications/applicationresources/assistance-schemes/educational-access-scheme
- Rawson, A. (2016). 'The Caring University in 2016: Practice, Partnership and Strategy with the Care Experienced Student, Action on Access eUpdate, July 2016.
- Smith, J. A., Flowers, P., & Larkin, M. (2009). *Interpretative Phenomenological Analysis: Theory, Method, Research*. London: Sage.
- The Centre for Social Justice. (2015). *Finding their feet: equipping care leavers to reach their full potential* (London: The Centre for Social Justice).
- Thoresen, S. H., & Liddiard, M. (2011). 'Failure of care in State care: in-case abuse and postcare homelessness', *Children Australia*, *36*(1), 4-11, doi: 10.1375/jcas.36.1.4
- Townsend, M. (2012). Are we making the grade? The education of children and young people in outof-home care. (Ashfield, New South Wales: Department of Family and Community Services).
- Universities Admission Centre (UAC). (2017). Educational Access Schemes (EAS). Retrieved 7 February 2017, from http://www.uac.edu.au/eas/
- Victorian Department of Human Services. (2007). The home-based care handbook. Melbourne: Department of Human Services.
- Victoria State Government. (2017). On Track. Retrieved 20 February 2017 from http://www.education.vic.gov.au/about/research/Pages/ontrack.aspx
- Victorian Tertiary Admissions Centre (VTAC). (2017). Special consideration (SEAS). Retrieved 7 February 2017, from https://blog.vtac.edu.au/category/special-consideration-seas/

Who Cares? Trust. (2016). Propel. Retrieved 31 January 2017 from http://propel.org.uk/

- Wilson, J. Z. & Golding, F. (2016). Muddling upwards: the expectation, unpredictable and strange on the path from care to high achievement in Victoria, Australia. In Philip Mendes and Pamela Snow (Eds), Young people transitioning from out-of-home care: international research, policy and practice (pp. 93-113). London: Palgrave Macmillan.
- Wolanin, T. R. (2005). *Higher education opportunities for foster youth: a primer for policymakers*, (Washington, DC: Institute for Higher Education Policy).
- Yosso, T. (2005). Whose culture has capital? A critical race theory discussion of community wealth. *Race, Ethnicity and Education, 8*(1), 69-91.

# Appendix: Interview guide

#### Participant information

- What is your age? Do you identify as Aboriginal or Torres Strait Islander?
- Where were you born?
- Where do you currently live and what is your living arrangement? What have your previous living arrangements been like, i.e. from care to shared or independent living? Can you describe your living arrangements particularly around your time of entry into university?
- Do you have children? If so how old are they?

#### Pathways to university

- Please describe your prior educational experience and your pathway into university.
   (i.e. when you finished secondary school, when you commenced university, and additional study you may have been enrolled in, such as TAFE). Do you believe your pathway into university differed from other students? If so, how?
- What barriers, if any, did you face accessing university?
- What encouraged you to attend university?
- What methods of applying for university were you aware of, and which ones did you use?
  - Were you aware of application processes which considered individual circumstances and if so did you apply? (for example SEAS/EAS, alternative entry schemes)
  - Were you aware of any equity scholarships and if so did you apply for any?
  - o Please describe any issues you experienced in the application and enrolment process.

#### **Experiences at university**

- What course are you enrolled in and how far along are you in the course? (i.e. full time/part time, online versus face-to-face attendance)
- What are your biggest challenges, if any, related to life as a university student?
- In what ways do you believe your experience at university might differ from students who have not experienced time in care?
  - In what ways might your care experience impact on your ability to achieve your best? How could this be addressed?
  - In what ways might your care experience give you an advantage over other students?
- What university services or programs do you use that you find the most beneficial?
  - What can the university do to help you excel? (i.e. are there additional university services or programs would you find beneficial?)
- What government or community services do you find most beneficial, and what additional services do you think are needed from them?
- If you were to receive financial assistance related to attending university, would you prefer targeted assistance for specific expenses such as book vouchers, travel tickets, or the provision of technology, or would you prefer direct financial aid? If targeted assistance, for what?
- What would be your response if you were offered a scholarship or bursary on the condition that you participated in defined university programs (such as regularly meeting with an academic advisor)?

#### **Future plans**

- Do you intend to complete your current course? If not, why not and what are your plans?
- Do you intend to continue onto postgraduate study either now or in the future? Why/Why not?
- What do you want to do once you have finished at university?

#### Recommendations

- Do you have any advice for young people still in out-of-home care, or people who have transitioned out of care regarding accessing and attending higher education?
- What advice would you provide to universities to help them encourage young people in care, and people who have transitioned out of care to apply for and enrol in higher education?
- What advice would you provide to universities to help them ensure care leavers excel once they enrol in university?
- How should universities reach young people still in out-of-home care?
- Would you be interested in getting together with other care leavers in groups?

Centre for Higher Education Equity and Diversity Research latrobe.edu.au/cheedr